

Remiern



# LoRaWAN Meets ML: A Survey on Enhancing Performance with Machine Learning

Arshad Farhad D and Jae-Young Pyun \* D

Wireless and Mobile Communication System Laboratory, Department of Information and Communication Engineering, Chosun University, Gwangju 61452, Republic of Korea; arshad@chosun.ac.kr \* Correspondence: jypyun@chosun.ac.kr

Abstract: The Internet of Things is rapidly growing with the demand for low-power, long-range wireless communication technologies. Long Range Wide Area Network (LoRaWAN) is one such technology that has gained significant attention in recent years due to its ability to provide long-range communication with low power consumption. One of the main issues in LoRaWAN is the efficient utilization of radio resources (e.g., spreading factor and transmission power) by the end devices. To solve the resource allocation issue, machine learning (ML) methods have been used to improve the LoRaWAN network performance. The primary aim of this survey paper is to study and examine the issue of resource management in LoRaWAN that has been resolved through state-of-the-art ML methods. Further, this survey presents the publicly available LoRaWAN frameworks that could be utilized for dataset collection, discusses the required features for efficient resource management with suggested ML methods, and highlights the existing publicly available datasets. The survey also explores and evaluates the Network Simulator-3-based ML frameworks that can be leveraged for efficient resource management. Finally, future recommendations regarding the applicability of the ML applications for resource management in LoRaWAN are illustrated, providing a comprehensive guide for researchers and practitioners interested in applying ML to improve the performance of the LoRaWAN network.

**Keywords:** LoRa; LoRaWAN; Internet of Things (IoT); machine learning (ML); resource management; spreading factor (SF); transmission power (TP); simulation; artificial intelligence; deep learning; reinforcement learning; dataset

## 1. Introduction

The Internet of Things (IoT) is a rapidly growing field that involves connecting a wide range of devices to the Internet to enable communication and data exchange between them. IoT enables seamless integration of the physical and digital worlds, revolutionizing various domains such as healthcare, transportation, agriculture, and industrial automation [1–4]. In IoT connectivity, several technologies have emerged to address the diverse requirements of IoT applications. These technologies comprise Long-Range Wide Area Networks (LoRaWAN), SigFox, Narrowband (NB)-IoT, Weightless, and Long Term Evolution for Machines (LTE-M) [5–8]. The key features of these IoT technologies are illustrated in Table 1. Sigfox offers a simple and low-cost deployment, while NB-IoT leverages the existing cellular infrastructure and provides higher data rates. The Weightless protocol provides flexibility and scalability, and LTE-M supports enhanced mobility and coverage. LoRaWAN [9] is among the leading low-power wide area network (LPWAN) technologies that have gained significant attention recently due to its ability to provide long-range communication with low power consumption. As a result, it has been extensively adopted by academia and industries for the IoT.



Citation: Farhad, A.; Pyun, J.-Y. LoRaWAN Meets ML: A Survey on Enhancing Performance with Machine Learning. *Sensors* **2023**, *23*, 6851. https://doi.org/10.3390/ s23156851

Academic Editor: Luca Leonardi

Received: 17 June 2023 Revised: 24 July 2023 Accepted: 27 July 2023 Published: 1 August 2023



**Copyright:** © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

Feature	LoRaWAN	Sigfox	NB-IoT	Weightless	LTE-M
Spectrum	ISM band (region-specific)	868 MHz, 915 MHz	1800 MHz, 2100 MHz	915 MHz, 2400 MHz	1800 MHz, 2100 MHz
Bandwidth	125, 250, and 500 kHz	100 Hz	Narrowband, typically 200 kHz	LTE bandwidth, typically up to several MHz	Typically several MHz
Modulation	Chirped spread spectrum	Ultra Narrowband)	GMSK	various (depends on the variant (GFSK)	OFDM
Payload size Data rate	Up to 243 bytes Up to 100 kbps	Up to 128 bytes Up to 100 bps	Up to 1600 bytes Up to 200 kbps	Up to 255 bytes Up to 1 Mbps	Up to 1500 bytes Up to 1 Mbps
Range [km]	Urban = 5, Rural = 20	Urban = 10, Rural = 40	Urban = 1, Rural = 10	Up to 10 km	Up to 10 km
Adaptive data rate	Yes	No	Yes	Yes	Yes
Energy consumption	Very low	Very low	Low to moderate	Low	Low to moderate
Mobility support	Yes (without ADR)	No	Limited	Limited	Limited
Localization	RSSI and TDoA [12,13]	No	No	Yes	Varies
Private network	Yes	No	Yes	Yes	Yes
Bidirectional communication	Yes	No	Yes	Yes	Yes
Deployment	Public	Closed	Public	Public	Public
Simulators [public]	Yes [14–30]	Yes [31–33]	Yes [34,35]	Not publicly available	Yes [35–38]

Table 1. Key features of widely adopted IoT technologies [10,11].

Long Range (LoRa) is the physical layer (PHY) primarily based on chirp spread spectrum (CSS) modulation, making it capable of achieving long-range and low power consumption [39]. LoRaWAN is the medium access control layer (MAC) responsible for efficiently managing communication between LoRa end devices (ED) and gateways (GW). In addition, LoRaWAN offers features such as adaptive data rate (ADR) for efficient resource management, bi-directional communication, and strong security, making it a robust and scalable solution for IoT deployments [40]. With their long range, ultra-low power consumption, and efficient network management, LoRa and LoRaWAN are revolutionizing the IoT landscape, empowering businesses and industries with seamless connectivity and enabling innovative IoT applications across various sectors [41].

#### 1.1. Existing Surveys on LoRa/LoRaWAN and Motivation

The specification of LoRa provides a detailed overview of the PHY and LoRaWAN features along with the ADR, retransmission procedures, and other features [42]. However, the decision-making of resource parameters configuration and optimal allocation to EDs [e.g., SF, bandwidth (BW), coding rate (CR), and transmission power (TP)] is left open to developers and academic researchers, allowing them to create and develop solutions for IoT applications.

In recent years, the LoRa/LoRaWAN has been surveyed in various aspects, such as ADR optimizations, mobility management, simulation tools, routing, security, etc., highlighting advantages, disadvantages, and future recommendations, as illustrated in Table 2. Table 2 presents the existing surveys and tutorials with the main focus and brief description of the topics covered on LoRa/LoRaWAN. In addition, the current surveys and tutorials in Table 2 are not focusing on the resource allocation issue addressed using machine learning (ML) methods for the LoRa/LoRaWAN. Therefore, this survey fills the stated gap by presenting a constructive and comprehensive review of the use of ML in LoRa/LoRaWAN.

# Table 2. Overview of surveys and tutorials on LoRa and LoRaWAN from 2018 to July 2023.

Ref.	Year	Main Focus of Survey	Brief Description of Main Topics Covered
[43]	2018	Pros and Cons of LoRaWAN	Compared existing solutions along with pros and cons and highlighted challenges and solutions.
[44]	2018	Security risks of LoRaWAN	Discussed and analyzed the impact and likelihood of each security threat in LoRaWAN.
[45]	2018	LoRa and its applications	Presented a comprehensive review regarding LoRa and its applications.
[46]	2019	advantages and disadvantages	Compared NB-IoT, LoRa, and Wi-Fi HaLow in terms of their main characteristics.
471	2019	Comparative study	Studied LoRaWAN, NB-IoT, LTE-M, and Sigfox and their use in WSN scenarios.
[48]	2019	Edge and Fog computing	Discussed LoRa-based edge and fog computing paradigms, highlighted pros and cons.
[49]	2019	LoRa/LoRaWAN challenges	Reviewed challenges of LoRa in scalability, capacity, and signal collision.
[50]	2019	Overview of LoRaWAN, DASH7, and NB-IoT	Reviewed the architectures and addressed the mobility management issues, and presented a comparative study of LoRaWAN_DASH7_and NB-IoT
[51]	2019	Overview LPWA technologies	Provided an overview of the existing solutions and identified key research challenges to be addressed in LPWA technologies
[52]	2019	LoRaWAN simulators	Reviewed several existing available simulators for LoRa/LoRaWAN along with design requirements and their limitations and how to improve simulators
[53]	2019	Security and energy	Overviewed LoRaWAN in terms of security and energy based on existing state-of-the-art
[54]	2019	Capacity of LoRaWAN	Studied the capacity of LoRaWAN in terms of ADR channels SF RF and ED density along with challenges
[01]	2017	cupacity of Ebharmin	and future research opportunities
[55]	2019	LoRaWAN simulators	Provided an overview of existing simulators for LoRa/LoRaWAN with requirements and limitations
[56,57]	2020	ADR optimization	Reviewed the existing ADR solutions, discussed the impact on the performance of LoRaWAN networks, iden-
[00,07]	2020	11D It op unitzation	tified challenges and future optimization techniques for improving the ADR.
[58]	2020	LoRa networking challenges	Investigated the challenges in terms of networking faced during deployment, presented recent solutions, and discussed opposisives considering practical large-scale deployment of LoPa networks
[59]	2020	Feasibility of adapting UDN	Carried the feasibility of adapting an ultra-dense network (UDN) within LoRaWAN and provided details of Mach L aRaWAN topology for UDN
[60]	2020	Visual data transmission	Evaluated existing techniques regarding the image transmission over LoRa networks, presented challenges,
[61]	2020	LoRaWAN mesh networks	Presented a review and comparative analysis on the classification of multihop communication solutions, dis-
[62]	2020	Security in LoRaWAN	Analyzed security issues and possible network attacks in LoRaWAN and presented countermeasures prevent-
[(0]	2020		ing LokaWAN from attacks.
[63]	2020	Confirment traffic in LaDaWAN	Discussed related approaches concerning multihop communication and routing protocols.
[64]	2020	Confirmed traffic in LokaWAN	Highlighted use cases, examined several aspects of confirmed traffic along with existing solutions.
[65] [66]	2020 2021	Use of ML in LoRa	Surveyed the general issues related to LoRaWAN, overviewed the ML solutions, and highlighted key future
[(7]	2021		research directions.
[07]	2021	Lokawan opumizations	resented existing solutions in five aspects: coexistence, resource anocation, MAC tayer, network plaining, and mobility support
[68]	2021	UAV-Based LoRa communication	Studied deployments of UAV-based LoRa network and reviewed systematically focusing on the communica-
[(0]	0001		tion setup and its performance.
[69]	2021	ADK enhancements	Reviewed the existing ADK solutions with regard to mobility.
[70]	2021	Comparative analysis LaBaWAN	Investigated routing approaches in multi-nop networks.
[/1]	2021	and NB-IoT	tives.
[72]	2021	Performance evaluation	Studied the factors affecting the capacity of the LoRa networks and its performance.
[73]	2021	Simulation tools	Presented the available simulation tools utilized for LoRaWAN performance assessment in ns-3.
[74]	2022	Resource allocation (e.g., SF)	Presented a concise overview of the traditional SF assignment methods to IoT end devices.
[75]	2022	LoRaWAN optimizations	Discussed various aspects, including bandwidth, modulation, data rate, coverage, link budget, payload, power officiancy occurity. ADR ontimizations and localization concorning LAPAWAN
[76]	2022	LoRa networking techniques	Surveyed the LoRa network techniques in LoRa (PHY layer), LoRaMAC layer (WAN), and application layers along with challenges and future trends
[77]	2022	LoRaWAN protocols	Provided an extensive survey on the existing LoRaWAN communication protocols focusing on the energy efficiency at both LoRa (PHY layer) and LoRaMAC layer (WAN)
[78]	2022	Energy efficiency	Surveyed the existing works on energy efficiency at LoRa and LoRaWAN.
[79.80]	2022	LoRa simulators	Presented a comparative study simulation tool for the simulation of LoRa/LoRaWAN networks.
[81]	2022	LoRaWAN security	Highlighted vulnerabilities and security attacks, discussed their systematic mitigation approaches.
[82]	2022	Recent advancement in LoRa	Reviewed LoRa concerning analysis, communication, security, and applications.
[83]	2023	Scalability in LoRaWAN	Discussed scalability challenges with existing state-of-the-art solutions to assist LoRaWAN deployment in mas- sive IoT networks
[84]	2023	Artificial Intelligence of Medical	Explored the current literature in the AIoMT, emphasized the powerful association between AI and IoT tech-
Our cur	N77	MI for recourse management	noncytes. This survey presents an in-depth review of recourse management issues with state of the art ML solutions.
Jui suive	- y	(e.g. SF. TP. BW)	discussing the LoRaWAN frameworks for dataset collection providing a constructive review on the ps-2-based
		(	ML frameworks, and presents future recommendations.

# 1.2. Methodology

We started with a systematic literature review methodology for this survey on improving LoRaWAN performance with ML, as illustrated in [85]. We first comprehensively searched the abstracts of all papers on IEEE Xplore, ACM, Elsevier, Wiley, and MDPI databases for LoRa/LoRaWAN and AI/ML/DL/RL. The search yielded approximately 148 papers, which include all aspects of LoRa/LoRaWAN resolved through ML techniques, as illustrated in Figure 1.



**Figure 1.** Number of published articles found in the existing literature for improving the performance of LoRaWAN using ML applications.

However, this survey only focuses on the performance improvement of LoRaWAN achieved using efficient resource management (e.g., channel allocation, SF, TP, and BW) through ML. Out of 148 papers, we found 36 papers dealing with resource management issues. This number also includes manually adding a small number of papers not found by our initial search, using backward reference searching and cross-citation techniques.

## 1.3. Scope and Contribution of the Survey

In contrast to published surveys and tutorials highlighted with a brief description of the main topics covered in Table 2, which present many characteristics or provide a comprehensive evaluation of the LoRa and LoRaWAN communication systems, where a comparison with other LWPAN technologies, potentials of both LoRa and LoRaWAN, ADR optimization/enhancement techniques, interference/collision mitigation, and available simulators for LoRa/LoRaWAN are the major topics, our survey mainly focuses on resource allocation issue addressed through ML for improving LoRaWAN performance.

The contribution of this survey, compared to the surveys and tutorials presented in Table 2, is as follows:

- We provide a systematic overview of the different areas of LoRaWAN performance where ML/DL/RL has been applied. We discuss the core LoRaWAN issues that can be addressed with ML/DL/RL and provide examples of how ML/DL/RL has been used to address these issues;
- 2. We discuss the publicly available LoRaWAN frameworks, which can potentially be applied for dataset collection. A comprehensive study has been carried out to highlight the best features for efficient resource allocation and the ML/DL/RL methods for improving LoRaWAN performance;
- 3. We extensively provide a discussion on the Network Simulator-3-based ML/DL/RL frameworks that could be utilized for efficient resource allocation with comprehensive scenarios;

 We identify open challenges in each area of LoRaWAN performance, discuss future research directions concerning resource allocation, and highlight potential benefits of ML for improving LoRaWAN performance.

# 1.4. Structure of the Survey

Section 2 highlights the core features of LoRa and LoRaWAN. Section 3 presents state-of-the-art methods for improving LoRaWAN performance using ML, DL, and RL. Section 4 elaborates on the existing LoRaWAN frameworks that could be utilized for dataset collection, discusses the required features, and highlights the best suitable ML methods for resource allocation concerning the features. Section 5 discusses the publicly available datasets utilized for various LoRaWAN deployments and IoT applications. Section 6 presents the existing publicly available ns-3-based ML frameworks and discusses how to utilize them for improving the performance of LoRaWAN. Section 7 presents a detailed discussion and highlights the potentials and limitations of the resource management ML methods applied on EDs and NS sides. Section 8 elaborates on the open research opportunities regarding efficient resource management, whereas Section 9 provides concluding remarks on this survey paper.

## 2. Core Features of LoRa and LoRaWAN

This section briefly presents the core features of LoRa and LoRaWAN.

## 2.1. LoRa-Long Range

LoRa [86] is a radio frequency (RF) modulation technology that defines the PHY layer features for long-range communications. LoRa is a proprietary PHY layer modulation based on CSS modulation to achieve long-range communication [87]. CSS is a subset of Direct-Sequence Spread Spectrum (DSSS), helping the GW to recover a weak signal and achieve high sensitivity, enabling increased coverage at a lower data rate (DR) [88,89]. In addition, LoRa utilizes five configurable resource parameters, i.e., SF, TP, CR, BW, and carrier frequency (CF), to fine-tune the link performance and energy consumption [56,74,79,90,91]. These configurable resource parameters of LoRa communication are discussed here.

## 2.1.1. Spreading Factor (SF)

The number of bits encoded in a symbol by LoRa is an adjustable resource parameter known as the SF. LoRa operates in six SFs (i.e.,  $SF_7 \sim SF_{12}$ ), utilized by the ED during uplink (UL) transmission. To transmit a UL packet, the EDs select a random channel using ALOHA channel access mechanism [92]. Furthermore, the choice of SF an ED utilizes during the communication plays a significant role for different reasons: a higher SF (e.g., 11, 12) complies with a high distance coverage; however, it indicates a low DR and high time-onair (ToA) [67,93,94]. For example, the ToA for a packet size of 51B and 1% duty cycle (DC), considering the EU region (i.e., 868 MHz), is illustrated in Table 3. Table 3 is computed using The Things Network (TTN) community network platform, where the TTN has utilized the LoRaWAN regional parameters [9], consisting of the duty-cycled limited transmissions to comply with the European Telecommunications Standards Institute (ETSI) regulations. In the EU region, the ETSI imposes DC limitations, where LoRaWAN complies with a maximum DC of 1%. The TTN uses a fair access policy (FAP) [95], allowing ED to send data to GW for at most 30 s of ToA and ten downlink messages (including acknowledgments for confirmed packets) per ED per 24 h [96,97]. Based on the TTN network,  $SF_{11}$  and  $SF_{12}$  are only allowed when ADR is enabled. Increasing the SF by one step doubles the ToA (for the same BW). It also indicates that a single transmission on  $SF_{10}$  takes more time than 6 on  $SF_{77}$ , or may need about the same ToA as 3 on  $SF_7$ ,  $SF_8$ , and  $SF_9$  combined. As a consequence of this behavior, the use of ADR or blind ADR is suggested for SF and TP adjustment [98–100].

Conditions	SF7	SF8	SF9	SF10	SF11	SF12
ToA [ms]	118.0	215.6	390.1	698.4	1478.7	2793.5
1% DC [s]	11.8	21.6	39.0	69.8	147.9	279.3
1% DC [msg/h]	305	167	92	51	24	12
FAP [avg $\cdot$ s]	339.9	620.8	1123.6	2011.3	4258.5	8045.2
FAP [avg/h]	10.6	5.8	3.2	1.8	0.8	0.4
FAP [msg/h]	254	139	76	42	20	10

**Table 3.** Time-on-air, duty cycle, and fair access policy (FAP) conditions at 125 kHz and packet size of 51B [101,102].

#### 2.1.2. Transmission Power (TP)

In LoRa, TP is an adjustable parameter with a step of 2, ranging from 2 to 14 dBm. TP is controlled by the ADR, implemented at the ED and NS sides to control the energy consumption of EDs [103,104].

## 2.1.3. Coding Rate (CR)

LoRa uses forward error correction (FEC) to improve the reliability of wireless transmissions. FEC adds redundant bits to the data, which can be used to correct errors occurring during transmission. The CR determines the redundancy added to the data. The smaller the CR, the more redundant bits are added, and the more reliable the transmission will be. However, a smaller CR will also increase the time it takes to transmit the data. The CR can be chosen among 4/5, 4/6, 4/7, and 4/8. The smallest CR, 4/8, provides the best reliability but takes the longest ToA to transmit the data. The largest CR, 4/5, provides the least reliability but sends the data the fastest. The choice of CR depends on the application that requires high reliability, such as industrial automation. A large CR should be used for applications that require fast data transmission, such as asset tracking.

#### 2.1.4. Carrier Frequency (CF)

CF in LoRaWAN is the frequency at which a LoRa ED transmits data toward GW. It is typically selected from a range of frequencies in a particular region. The CF affects the capacity and power consumption of ED. For example, the LoRa CF can be programmed in steps of 61 Hz between 137 MHz to 1020 MHz. However, depending on the particular LoRa chip, this range may be limited to 860 MHz to 1020 MHz [103]. LoRa supports different ISM frequencies (in MHz), namely EU863-870 (Europe), US902-928 (North America), EU433 (Asia), CN470-510, CN779-787 (China), AU915-928 (Australia), KR920-923 (Korea), and IN865-867 (India) [9].

## 2.1.5. Bandwidth (BW)

LoRa operates in three BW: 125, 250, and 500 kHz. The BW is determined by the regional parameters, as specified in the LoRaWAN specifications [9]. A LoRa-modulated signal comprises 2<sup>SF</sup> chips spread over the available BW. The SF parameter controls the spreading BW and the signal sensitivity to noise. A larger SF value results in a wider spreading of BW and lower sensitivity to noise. However, it reduces the DR.

#### 2.2. Long Range Wide Area Network (LoRaWAN)

LoRaWAN defines MAC layer features as consisting of a star-of-stars topology comprising many EDs, GW, NS, and application servers, as shown in Figure 2. The EDs in LoRaWAN network are classified as *Class A*, *Class B*, and *Class C* [105]. *Class A* EDs are battery-powered and consume ultra-low energy. These EDs are bi-directional and receive acknowledgment (ACK) from NS with two available receive windows (*RXs*). *Class B* EDs are also battery-powered and provide bi-directional communication. These EDs support unicast and multicast transmission, though they have more *RXs* and are synchronized with a beacon frame transmitted by the GW after a certain time. Finally, *Class C* EDs use more power and listen all the time, excluding the transmission time. Among these EDs classes, *Class A* EDs deal with sensors and are implemented in IoT applications, owing to their energy efficiency and bi-directional communications [79]. Furthermore, LoRaWAN supports two communication modes: confirmed and unconfirmed.



Figure 2. LoRaWAN network.

## 2.2.1. Confirmed Mode

In LoRaWAN, the ED initiates data transmission with an SF and TP. The SF and TP are allocated by the NS using the ADR mechanism, as defined in [99,106–113]. The ADR determines the values of SF and TP based on the highest SNR value of the last 20 packets received at the NS. The NS reduces the SF and increases or decreases the value of TP by 2 to reduce energy consumption. However, the newly adapted SF and TP might not successfully deliver the packet to the NS. Therefore, in confirmed mode, the ED utilizes a recovery ADR based on the retransmission procedure on the ED side. When the retransmission counter is a multiple of two, the ED increases its SF, and a TP of 14 dBm is adopted at the time of packet transmission [114]. It increases the chances of successfully delivering a packet to the NS with increased energy consumption costs.

#### 2.2.2. Unconfirmed Mode

The unconfirmed mode does not require a downlink ACK from the NS. However, to determine the connectivity loss between the ED and GW, the ED enables ADR ACK bit by sending a MAC command *ADRACKReq* in the LoRa frame header (FHDR) after 64 (default) UL packets [9]. In such a case, the NS must send an ACK, but not immediately. Furthermore, LoRaWAN utilizes ADR for SF and TP management in confirmed and unconfirmed modes.

## 3. LoRaWAN Meets ML

This section mainly focuses on existing ML methods applied to LoRa and LoRaWAN for efficient resource management (e.g., SF, TP, BW, and CR) for improving LoRaWAN network performance and efficiency. In the remainder of this section, we present the existing ML, DL, and Reinforcement Learning (RL) methods applied to LoRaWAN.

## 3.1. Improving LoRaWAN Performance Using ML

Here, we identify the need for ML and present state-of-the-art methods for enhancing LoRaWAN performance through efficient resource management. These ML methods applied for improving the performance of LoRa and LoRaWAN are shown in Table 4.

#### 3.1.1. Need for Machine Learning

ML is a rapidly growing field with many applications, including wireless communications [115–119]. ML can be utilized to improve the performance, efficiency, and security of wireless networks [120,121]. However, ML is applied in a mathematical model deficit and algorithm deficit cases in IoT scenarios [122]. In LoRaWAN, resource management decision-making is left open to developers and researchers, allowing them to develop intelligent solutions for demanding IoT applications. One approach is to utilize ML for resource management, revolutionizing the optimization of SF, TP, BW, and other important parameters. ML algorithms empower LoRaWAN networks to dynamically allocate resources, predict network traffic, mitigate interference, and optimize energy consumption, thereby enhancing network capacity, reliability, and battery life. With ML-driven insights, operators can proactively plan network expansions, ensure the quality of service (QoS), and achieve self-optimizing networks that autonomously adapt to changing conditions [123,124]. This cutting-edge technology releases the full potential of LoRaWAN, transforming it into an intelligent, adaptive, and efficient IoT infrastructure for a wide range of applications.

#### 3.1.2. Machine Learning: The State-of-the-Art

The existing state-of-the-art ML methods can be classified into supervised and unsupervised.

#### Supervised Approaches

A load-balancing method for dense heterogeneous IoT networks, such as smart city scenarios, was proposed in [125]. The dataset was gathered from a TTN Mapper (mapping the coverage of TTN GWs based on user data) [126] of frequency, DR, latitude, longitude, RSSI, and SNR. These features describe the successful UL packet transmission from ED to the GW. The authors trained different ML techniques, such as Multiple Linear Regression (MLR), Gaussian Naive Bayes (GNB), Linear Discriminant Analysis (LDA), Quadratic Discriminant Analysis (QDA), Decision Tree (DT), Random Forest (RF), Extremely Randomized Trees (ET), and Voting (Ensemble Learning). The classifiers were applied to an urban IoT network, where the simulation results showed an improved packet success ratio (PSR) and reduced energy consumption of a LoRaWAN network.

In [127], an SF allocation scheme using a support vector machine (SVM) and DT to resolve the collision issue in the LoRaWAN network has been proposed. The training dataset was generated using Simulator for LoRa SimLoRaSF [128] (SF), a custom simulator designed for LoRaWAN using Python. The dataset contains the X and Y coordinates of the ED along with successful SF. The input is labeled as successful (if the packet was successfully received at the GW), interfered (when a packet was unsuccessful due to Co-SF interference), and under sensitivity (when a packet is arriving under the required sensitivity threshold at a specific SF). The SVM and DT classifiers are trained for optimal SF allocation. Their simulation results showed that the SVM and DT could efficiently classify the SF, improving the PSR and transmission energy consumption compared to the random SF allocation method. However, the SimLoRaSF does not consider the downlink communication, which is an essential part of the LoRaWAN.

The authors in [129] solve the resource classification problem (e.g., TP) for static EDs in LoRaWAN through various ML techniques, such as RF, SVM, logistic regression (LR), K-nearest neighbor (KNN), LDA, and GNB. The authors used LoRaSim [30,130,131] network simulator for dataset collection, which is designed for LoRaWAN IoT networks based on Python. The ML algorithms were trained on data from previous packet transmissions, such as SF, CR, Nakagami path loss, and the distance between the ED and GW. As a result, every combination of SF and CR pairs has one optimal TP associated with it. The dataset was split into training and testing by 70% and 30%, respectively. From classification results, it was observed that the RF method achieved the highest accuracy of 92.96% compared to other ML techniques. Their simulation results revealed that suitable TP classification leads to a higher PSR than other non-ML methods, such as ADR.

The authors in [132] proposed a combined path loss and shadowing (CPLS) technique, where ML methods such as LR, SVM, RF, and Artificial Neural Network (ANN) were trained on RSSI, ToA, SF, and SNR. The dataset was collected through a testbed utilizing four static EDs in a line-of-sight (LoS) scenario. After removing the outliers and wrong data collected from the sensors, they divided the dataset into training (80%) and testing (20%). To this end, the authors suggested an enhanced ADR for SF and TP allocation to static EDs. The ML methods were evaluated with root-mean-square error (RMSE), achieving up to 1.566 dB and  $R^2$  up to 0.94. The enhanced ADR results revealed reduced energy consumption by 43% compared to the ADR of LoRaWAN.

The paper [133] proposed an ML approach based on a learning-automata mechanism to extend the lifetime of IoT EDs utilized for forest monitoring. Their proposed approach selects the most energy-efficient ED to act as a cluster head. The simulation results showed that the proposed learning-automata mechanism increased the lifetime of IoT EDs by up to 6.7 times. Their proposed approach has empirically proven that the learning automaton consistently converges on the most suitable ED to serve as the cluster head based on energy consumption metrics, proving their proposed approach is scalable and can easily be adapted to different IoT applications.

#### Unsupervised Approaches

One of the challenges of LoRaWAN is the collision occurring between the packets of EDs when transmitted at the same time with the same SF over the same channel. A K-means clustering algorithm was proposed for reducing collision probabilities [134]. Their proposed K-means are responsible for grouping EDs with similar traffic patterns. EDs belonging to different groups have different priorities, and the channel access for the UL packet transmission is determined by the given priority. Their method showed improved performance in terms of collision probability for class *A* and *B* devices. Their proposed method is simple, scalable, and efficient for reducing collision probabilities in LoRaWAN. Similarly, ED profiling using K-means was utilized in [135] to predict the behavior of Lo-RaWAN traffic. The authors grouped the EDs using the same SF and packet size and trained DT and Long Short-Term Memory (LSTM) using unsupervised traffic pattern classification methods. Their simulation results showed improved performance in terms of PSR by reducing the impact of interference. However, in a dynamic LoRaWAN network, ED profiling can be time-consuming because resources (e.g., SF and TP) are application-dependent. Therefore, their proposed method is applicable to static LoRaWAN network environments.

A learning-automaton-based ML approach was proposed on the NS in [136] to select between two MAC protocols: TDMA and Slotted ALOHA. The selection of each MAC protocol depends on the network traffic load. If the network traffic load is high, TDMA is chosen for packet transmission to avoid collisions. If the network traffic load is low, Slotted ALOHA is utilized for communication to reduce packet delay. The learning automaton adapts the MAC protocol selection based on the feedback received from the environment. The proposed learning automaton-based ML approach has been evaluated in simulation, showing improved performance in the presence of event traffic.

Ref.	Year	ML Model(s)	ML Approach	Features	Deployment Platform	Dataset Tool	Application(s)
[125]	2018	MLR, GNB, LDA, QDA, DT, RF, ET, Voting	Supervised	RSSI, SNR	Python tool	TTN Mapper [126]	Smart city
[127]	2019	SVM, DT	Supervised	X and Y coordinates, SF	SimLoRaSF simula- tor [128]	SimLoRaSF sim- ulator [128]	×
[129]	2023	RF, SVM, LR, KNN, LDA, GNB	Supervised	SF, CR, path loss, and distance	LoRaSim	LoRaSim	Smart parking
[132]	2023	LR, SVM, RF, ANN	Supervised	RSSI, ToA, SF, SNR	x	Testbed	X
[133]	2023	Learning-automata	Supervised	RSSI, ToA, SF, SNR	×	Testbed	Forest monitor- ing
[134]	2019	K-means	Unsupervised	Traffic patterns and priority	x	Simulation	×
[135]	2020	DT, LSTM	Unsupervised	SF and packet size	x	Tested	Water metering
[136]	2022	Learning-automata	Unsupervised	Traffic pattern	x	Simulation	Environmental monitoring
[136]	2022	AR, TFT	Unsupervised	Time-stamp, SNR, duty cycle, number of transitions, SF, fre- quency, ADR, successful trans- mission, and failed transmis- sion	×	ns-3 simulator	Smart home/city

Table 4. ML methods applied for improving the performance of LoRa and LoRaWAN.

*X* = not mentioned in the referenced paper.

The authors in [137] proposed two AI methods, an autoregressor (AR) model and a temporal fusion transformer (TFT) model, for classifying and detecting LoRaWAN traffic to optimize the LoRaWAN network performance. The authors collected a dataset using the ns-3 simulator, where the EDs are placed in an 8 km circle with a centered GW. They considered all possible criteria as features for EDs, including the inter packets time, SNR, DC, number of UL transmissions, SF, frequency, ADR, successful transmission at the GW, and failed transmission. The simulation was executed for 365 days, resulting in a 100 GB database. The TFT method was utilized to forecast the behavior of the LoRaWAN network, and the AR method to detect the surge in traffic with overall classification precision between 94.50 and 99%. However, the suggested AI methods have not been utilized in the ns-3 for online testing.

## 3.2. Improving LoRaWAN Performance Using DL

We begin by highlighting the importance of DL in the LoRaWAN network. We then delve into the cutting-edge DL methods currently employed for solving the resource management issue to enhance the performance of LoRaWAN networks. These DL methods applied for improving the performance of LoRa and LoRaWAN are illustrated in Table 5.

## 3.2.1. Need for Deep Learning

DL can improve the performance of LoRaWAN networks by optimizing resource parameters, predicting network traffic, mitigating inter- and intra-interferences, and optimizing energy consumption [138–148]. Furthermore, to dynamically allocate resources to EDs, such as BW, SF, and TP, a DL method can be trained on a large dataset generated using simulation tools (e.g., ns-3 or Matlab) or testbeds. Finally, the trained DL method can be deployed on EDs or network servers (ns-3 or testbed deployments) for efficient resource allocation, improving the performance of the LoRaWAN network.

#### 3.2.2. Deep Learning: The State-of-the-Art

An Extended Kalman Filter (EKF)-based LSTM method based on regression method for predicting collision in LoRaWAN network was proposed in [141]. They generated the dataset for a number of collisions for each 20 min interval using the *LoRaSim* simulator [131]. For training the LSTM, the dataset was split into 70% and 30% into training and testing and scaled to [0, 1]. They utilized the pre-trained LSTM along with EKF for collision analysis. Their results showed an improved RMSE of 0.9863 compared to other approaches, such as Gated Recurrent Unit (GRU) and Recurrent Neural Network (RNN). Hence, with low RMSE, the LSTM-EKF has reduced the number of collisions in online simulation and yielded better performance. The collision in the LoRa network is directly linked with the SF; hence, SF has not been considered for adaptive configuration. As a result, their proposed LSTM-EKF can lead to underperformance when utilized in a dynamic LoRaWAN network.

The study in [146] proposed DeepLoRa, an environment-aware path loss model, utilizing satellite photos to categorize a land cover using Bi-LSTM accurately. In DeepLoRa, first, each pixel of the picture map was class-labeled, which divided the land cover into two classes, non-line of sight (NLoS, buildings, trees) and LoS (no attenuation), to reflect the actual land-cover type. Second, they divided LoRa lines from an ED to GW into identically sized micro-links. Each was then embedded into a different sequence element based on a land cover map. To determine the Estimated Signal Power (ESP) received by GW, the model integrated the sequences with specific input parameters and anticipated related path loss. For all land cover categories, the accuracy of land cover classification was 97.4% (which can be regarded as a true environment reflection). Furthermore, DeepLoRa performed at least 50% better than other models.

A neural-enhanced LoRa demodulation method (NELoRa) was proposed in [143]. A Deep Neural Network (DNN) was trained on a spectrogram of amplitude and phase. The authors conducted indoor and outdoor experiments, and three metrics, i.e., Symbol Error Rate (SER), SNR Gains, and Battery Life Gain (BLG), were used for the performance evaluation of NELoRa. The evaluations showed that NELoRa could obtain much lower SERs. As a result, the proposed NELoRa brought significant gains in the SNR thresholds compared to the dechirp process, and the highest SNR gain observed was 5.94 dB under *SF*<sub>7</sub> and 500 kHz of BW. NELoRa resulted in consistent and higher BLG of 27% compared to the baseline.

The paper proposed a DL method [149] for managing the transmission interval of IoT devices in LoRa networks by utilizing Intel Berkeley Research Lab Data [150]. Using an autoencoder, the proposed method first clusters the IoT EDs concerning their data patterns. Second, a local LSTM prediction model is trained using the Intel lab IoT dataset for each cluster to predict the next transmission interval of each ED involved in communication. The Monte Carlo simulation with the Intel lab IoT dataset showed 31% improved scalability.

The paper [147] proposed a DL method for joint collision detection and resource management (e.g., SF). The proposed work utilizes two DL methods: fully connected neural networks (FCNNs) for collision detection and CNN for SF management. The dataset used to train the DL methods was generated using the *SimLoRaSF* simulator [127], containing the X and Y coordinates of the ED along with SF. The results showed improved prediction accuracy and energy consumption compared to traditional ML methods such as SVM, DT, and RF.

Recently, the authors in [148] proposed an AI framework for SF classification using GRU. To train the GRU model, the dataset was generated in ns-3 and labeled in two ways:

- 1. *Group-based SF labeling:* the 6 UL packets, represented with  $UL_1, UL_2, ..., UL_6$  transmitted with  $SF_7$  to  $SF_{12}$  were organized into one group (*g*). From each *g*, the lowest SF was chosen among successful ACK receptions;
- 2. *Input sequence labeled with SF:* once the SF labeling of each *g* was completed, the authors generated an input sequence of 20 groups with a corresponding labeled SF.

Their proposed AI framework is comprised of two modes: offline and online. The GRU model is trained based on one-time generated data in the offline mode. After training, a pretrained (i.e., inference model) is utilized in the online mode, which yields the best SF during real-time simulation. The inference model was utilized on the ED side, where the new data with 20 UL sequences for determining the best SF was generated. Once the UL sequence reached a size of 20, the input was fed into the inference model to predict a suitable SF for the next UL packet transmission. Simulation results showed improved PSR compared to the typical ADR approach. However, the GRU model with two layers comprised 990.71k parameters (i.e., space complexity) and 249.67 Mega Floating-point Operations Per Second (MMac FLOPs, representing time complexity); the proposed model deployed on LoRa devices would be computationally costly. To address this issue, the authors in [114] proposed a DNN with reduced computational 13.54 MMac FLOPs and space complexity 52.9k parameters with the use of six sequences of UL packets during the online mode to lower the convergence period and energy consumption.

#### 3.3. Improving LoRaWAN Performance Using RL

Here, we discuss the advanced RL methods utilized to address the resource management challenge and further enhance the overall performance of LoRaWAN networks. The cutting-edge RL techniques are designed to intelligently manage network resources like BW, SF, and TP to maximize efficiency and deliver optimal results. These RL methods improving the performance of LoRaWAN are highlighted in Table 6.

Ref.	Year	DL Model(s)	DL Approach	Features	Deployment Platform	Dataset Tool	Application(s)
[141]	2020	LSTM, LSTM-EKF	Regression	Collisions	LoRaSim	LoRaSim	Smart city
[146]	2021	Bi-LSTM	Supervised	Pathloss	Testbed	Testbed	Localization
[143]	2021	DNN	Supervised	RSSI and SNR	Testbed	Testbed	Smart parking
[149]	2021	Autoencoder, LSTM	Supervised	Sensory data (temperature, humidity, light, voltage)	Monte Carlo simulation	Intel Lab Data [150]	x
[147]	2022	SVM, DT, FCNN, CNN, RF	Supervised	X and Y coordinates of the ED, SF	SimLoRaSF simula- tor [127,128]	SimLoRaSF simula- tor [127,128]	x
[148]	2022	GRU	Supervised	ED position, SNR, received power	LoRaWAN ns-3 [151]	LoRaWAN ns-3 [151]	Metering
[114]	2023	DNN, LSTM, GRU, SVM	Supervised	ED position, SNR, received power	LoRaWAN ns-3 [151]	LoRaWAN ns-3 [151]	Pet-tracking and metering

Table 5. DL methods applied for improving the performance of LoRa and LoRaWAN.

*X* = not mentioned in the referenced paper.

# 3.3.1. Need for Reinforcement Learning

RL approach has several advantages over traditional methods of optimizing Lo-RaWAN networks [129,152–155].

- RL is a data-driven approach, learning rules and policies from experience by interacting with the network and observing the results;
- RL is a dynamic approach, adapting itself to environmental changes. LoRaWAN
  networks are constantly changing owing to ED mobility and the underlying propagation environment. As a result, RL can be used to learn how to optimize the network
  parameters for these changes, ensuring that the network remains reliable and efficient;
- 3. RL is a scalable approach; thereby, it can be used to optimize large and complex networks.

RL has been utilized in different IoT applications, such as robotics, game-playing, network policy control, and resource optimization [156–159]. For example, in LoRaWAN, RL could be used to optimize the resources (e.g., SF and TP) by training agents (i.e., EDs) for making decisions to maximize the overall network efficiency and minimize interference and energy consumption [160,161].

#### 3.3.2. Reinforcement Learning: The State-of-the-Art

Paper [162] proposed an RL approach for optimizing and updating LoRa communication parameters. The authors mathematically modeled the average per-node throughput of LoRaWAN networks by considering the heterogeneity of IoT deployments. The authors utilized the RL method to derive optimal disseminating policies by aiming to maximize the accumulated average per-node throughput. The authors compared their approach to the LoRaWAN ADR mechanism. The authors showed that their approach achieved a remarkable increase in the accumulated average per-node throughput of 147%.

Paper [163] proposed a novel method for resource allocation in LoRaWAN networks. The method used Q-learning, an RL technique, to learn the optimal resource allocation policy for each ED in the network. In the proposed method, the GW acts as an agent of Q-learning, where the Q-reward is based on the weighted sum of the number of successfully received packets in the proposed method. The Q-learning method was evaluated using simulation, and it has improved the average PSR by about 20% compared to a random resource allocation scheme.

Ref.	Year	RL Model(s)	Reward Feature(s)	Platform	Application(s)
[162]	2019	Evolution strategies (ES)	Channel conditions	Simpy [164]	x
		algorithm		17.1	
[163]	2019	UČB	Q-learning	x	Gas and water meters
[165]	2019	UCB	ACK	Matlab	Smart metering
[166]	2020	DQN	PDR, ToA, and power usage	x	x
[167]	2020	DQN	Network reliability and the power effi-	Simulation	×
			ciency		
[168]	2020	DQN	Mobility, channel conditions, traffic load	Simulation	x
[159]	2021	Q-learning	x	x	x
[169]	2021	RL	SF Sensitivity	LoRa-MAB [21,170]	Monitoring
[154]	2021	RL-ADR	ToA and energy consumption	ns-3	x
[152]	2021	RL	DER and SNR	LoRaEnergySim [171]	Industrial monitoring
[153]	2021	DRL	Energy cost	Monte Carlo simulations	Real-time application
[172]	2022	MAB	ACK	LoRa-MAB [21,170]	Metering
[155]	2022	DRL	Number of ED, reliability, energy effi-	Testbed	x
			ciency		
[173]	2022	DRL	Collision rate and the packet loss rate of	LoRaSim simulator [131]	×
			EDs		
[174]	2023	Q-Learning, Boltzmann ex-	Success probability	MULANE [175]	×
		ploration algorithm			
[176]	2023	Multi-arm bandit	Energy	ns-3 module [151]	×
[177]	2023	DRL	Collision or packet lost	LoRaSim simulator [131]	×
[178]	2023	Multi-agent regression	Number of EDs, distribution of EDs,	Simulation	×
		model	the traffic pattern of EDs		

Tabl	e 6.	RL	methods	s for	imp	oroving	the	performance	of	LoRa	and	LoRa	NA	N.
------	------	----	---------	-------	-----	---------	-----	-------------	----	------	-----	------	----	----

*X* = not mentioned in the referenced paper.

To improve the PSR, the authors in [165] proposed a retransmission method based on an Upper Confidence Bound (UCB) algorithm that is used to solve the multi-armed bandit problem. In [165], first, the ED retransmits a packet with random channel selection and learns the quality of each channel based on a positive ACK reception. Then, after learning the best channel for retransmission, the EDs can retransmit on the highest-rewarded channel. Hence, improving the PSR. Their results showed improved PSR compared to random channel selection schemes.

A deep RL (DRL) method was proposed for the dynamic adjustment of SF and TP to mitigate the collision problem in LoRaWAN [166]. The authors considered the PSR, ToA, and TP of an ED as a reward function. To mitigate the collision behavior of the LoRaWAN network, a deep Q-network (DQN) was deployed at the GW for SF and TP management. As a result, their proposed method improved the PSR by 500% under 100 EDs deployed in a 4.5 km region.

The authors in [167] proposed a multi-agent Q-learning algorithm to dynamically allocate TP and SF to EDs during UL packet transmission in a LoRa network. In a LoRa multi-agent system, each agent represents one LoRa ED, where the ED works together with the environment to determine the best TP and SF allocation policies for every UL transmission. The simulation results demonstrated that the suggested algorithm could greatly enhance the energy efficiency and reliability of LoRa networks.

The authors in [168] proposed a DRL method called LoRaDRL-based on DQN for intelligent resource allocation in dense LoRa networks. The primary aim of the proposed LoRaDRL approach is to learn the optimal policy for allocating channels to LoRa EDs. LoRaDRL assigns resources to the EDs by considering the mobility of EDs, the channel conditions, and the traffic load in the network. LoRaDRL results showed improved PDR under dense deployments and mobile EDs compared to the state-of-the-art resource allocation algorithms, such as LoRaSim [131] and LoRa-MAB [21,170].

An RL algorithm was introduced in [169] for SF and TP optimization to improve energy consumption and PSR. The SF and TP are optimized based on the required level of SF sensitivity threshold of the GW. They implemented the RL algorithm in the *LoRa–MAB* simulator [21,170], which is based on Exponential Weights for Exploration and Exploitation (EXP3). The results showed an improved PSR and energy consumption compared to the LoRa-MAB algorithm [21]. Similarly, an RL-ADR was proposed in [154] to optimize the SF allocation. The agent is trained using the LoRaWAN ns-3 module, considering the variations in the SNR behavior. In addition, the reward function is based on the ToA and energy consumption. Both ToA and energy consumption are measured from each last received packet at the NS. The RL agent learns the efficient SF allocation policies based on the ToA and energy consumption. The proposed RL-ADR is evaluated in comparison with the traditional ADR mechanism of LoRaWAN, where it showed improved energy consumption owing to the fast response of the trained RL agent for efficient SF allocation during communication by the NS. Furthermore, [152] also proposed an RL-based optimized ADR for SF and TP allocation to EDs. The authors of [152] utilized *LoRaEnergySim* simulator [171], where the states are data extraction rate (DER) and SNR pairs. Moreover, [152] considers SF and TP pair as actions. The results of [152] showed improved DER compared to the traditional ADR mechanism. Another DRL approach has been proposed for optimal channel and SF assignment to EDs in LoRa networks [153]. Their proposed DRL approach utilizes a DQN to learn the optimal policy for channel and SF allocation. They trained the DQN on a historical data dataset to minimize the grid power consumption while satisfying the QoS requirements of the EDs. The proposed approach is evaluated in Monte Carlo and RL simulations and showed improved performance in assigning a suitable channel to EDs, thereby lowering the energy cost.

The authors of [172] proposed the MIX-MAB algorithm for suitable transmission parameters (i.e., SF) allocation to EDs. In MIX-MAB, LoRa EDs interact with the environment, including GWs, to learn the best actions based on the successful reception of ACK messages. In MIX-MAB, an ED always initiates a UL packet transmission towards GW using an SF. In return, the ED receives an ACK upon the successful reception of the packet on the NS. When the ED receives an ACK, it assigns a reward to that successful SF. As a result, this SF is used for the next UL packet transmission. The MIX-MAB was evaluated in *LoRa–MAB* simulator [21,170] with one GW located at the center of a disc-shaped cell with a radius of 4.5 km, where 100 LoRa EDs were uniformly distributed, each ED transmits 15 packets/hour. The simulation results showed improved convergence time and PSR compared to the LoRa-MAB algorithm.

The authors of [155] proposed a multi-agent DRL ADR mechanism at the NS side for efficient SF and TP allocation to EDs. The proposed multi-agent DRL ADR mechanism consists of three independent DRL algorithms, one for each slice, replacing the traditional LoRaWAN ADR mechanism for assigning TP and SF to EDs. The SF and TP are allocated to EDs based on the rewards such as the number of ED, reliability, and energy efficiency. Their proposed multi-agent DRL ADR mechanism showed improved energy consumption compared to the traditional ADR. Another approach based on DRL for optimizing the SF allocation is studied to improve the GW capacity of the LoRa network [173]. The proposed approach utilizes a DQN to learn the optimal SF assignment policy for a given network state. The reward is computed based on the collision rate and the packet loss rate of EDs. The authors of [131] performed simulation using the *LoRaSim* simulator, where the results showed a reduced collision rate by up to 30% compared to the existing Min-airtime and Min-distance based SF allocation approaches [130].

An algorithm called Low-Power LP-MAB (MAB) [179] was designed to configure the transmission parameters (e.g., SF) of ED in a centralized manner to improve energy consumption and PSR. The LP-MAB algorithm works on the NS side by interacting with the ED. The NS transmits ACK upon successful packet reception to the ED, where the ED learns the best SF for the subsequent UL packet transmission based on the received ACK for the previous successful communication on a particular SF. As a result, the simulation results of LP-MAB outperform other approaches in terms of energy consumption and PSR.

A Q-learning approach was proposed in [174], known as the score table-based evaluation and parameters surfing (STEPS) approach. STEPS is responsible for dynamically allocating the required SF for UL transmission based on the success probability of the packet and score table. The simulations were conducted using *MULANE* [175] simulator for different EDs (e.g., 50, 100, 250, 500, 600, and 750). Initially, during the deployment phase, all EDs utilize the same SF. Their results revealed that their proposed STEPS approach could reduce energy consumption by 24% to 27%. Furthermore, it was realized that this achievement was possible due to a reduction in collisions. In other experiments regarding bi-directional communication, their proposed STEPS approach enhanced the network throughput by 18% in a smaller network, while 33% in a relatively larger network compared to ADR, BADR, and LoRaMAB [170].

The paper [176] proposed a lightweight RL approach for appropriate SF allocation to EDs in a LoRaWAN network. The lightweight RL approach utilizes MAB to learn the tradeoff between energy consumption and DR. To ensure feasibility, the authors have integrated explicit MAC commands into their proposed method and implemented them in the ns-3 module [151]. Their extensive simulation results showed that their lightweight RL approach outperforms the traditional ADR in single and multi-GW scenarios regarding PSR and energy consumption, owing to learning the optimal SF for each ED in a given environment.

The authors of [177] proposed an SF redistribution method under limited network resources to improve the ED capacity of the LoRa GW. Their proposed method uses a DRL technique to learn the optimal SF allocation for each node, minimizing the collision rate and energy consumption. Simulation results using *LoRaSim* [131] showed improved capacity.

Paper [178] proposed a multi-agent regression model to improve network planning in time-slotted communications for LoRaWAN. The proposed agent is based on multi-output regression responsible for predicting the network scalability for a given set of joining EDs. The dataset used for training the agent was generated using a series of simulations, considering the features such as the number of EDs, the distribution of EDs, the traffic pattern of EDs, and the channel conditions. The agent utilizes the dataset to train the multi-output regression model. Once the model is trained, the agent predicts the network scalability for a given joining EDs. The simulation results revealed a 3% reduction in the mean absolute error, indicating that the agent can make accurate predictions.

#### 4. Simulators and Frameworks for Dataset Collection

In the existing literature, few works have surveyed the publicly available LoRaWAN network simulators [52,55,73,79,80,180]. Therefore, this section highlights a few LoRaWAN frameworks that have been utilized or can be used for dataset collection to resolve collision and resource management issues in the LoRaWAN network. Furthermore, Table 7 illustrates a comparison of dataset collection frameworks and applicable ML methods based on suggested features for resource classification.

#### 4.1. LoRaSim: LoRa Simulator

LoRaSim is based on Python, designed to simulate LoRaWAN collision behavior, and mainly consists of four configurations: (1) it simulates a single GW, (2) supports up to 24 GWs, (3) simulates EDs and GWs with a directional antenna, (4) and comprised of multiple networks [30,130,131].

LoRaSim [131] can be used to study the performance of different LoRaWAN network configurations and to evaluate the impact of interference. The interference model in LoRaSim is comprehensive, considering both co-SF and inter-SF interference. Co-SF interference occurs when two or more packets are sent on the same SF. Inter-SF interference occurs when a packet is sent on a different SF than another packet.

On the one hand, a packet is received correctly if it satisfies three thresholds: the minimum co-SF, the minimum inter-SF, and the minimum SNR. On the other hand, a packet is lost only if the overlap of packets is in the time-critical region of the considered packet. The time-critical region is the part of the packet most important for correct reception.

	Dataset Collection Frameworks				ML Techniques with Required Features for Learning		
Features	LoRaSim	SimLoRaSF	LoRaWAN- Sim	ns-3 Module	Applicable ML Techniques	FeaturesforResourceClassification (e.g., SF, TP)	
Simulation plat- form	Python	Python	Python	ns-3	x	×	
Frequency region	EU-868	EU-868	EU-868	EU-868	×	x	
Device type	Α	Α	Α	Α	×	X	
ADR	x	x	×	$\checkmark$	RL [21,154,170]	ToA, energy consumption, ACK, PSR [21,154,170]	
Propagation loss	log- distance	log- distance	Okumura Hata	log- distance	RF, DT, SVM, DNN	RSSI, SNR, distance, frequency,	
Energy consump- tion model	✓	√	$\checkmark$	√	SVM, DNN, Ensemble, Naive Bayes, DT	Payload Size, DR, TP, SNR, chan- nel occupancy, distance from GW	
Mobility environ- ment	X	X	x	✓	SVM, KNN, LSTM, DNN, RL, Hybrid	RSSI, speed, acceleration, location, trajectory, SNR, time-stamp	
Buildings environ- ment	x	x	x	~	ANN, SVM, GPs, DT	Height, density, material, obstruc- tions, RSSI, NLoS conditions	
Interference model	✓	$\checkmark$	$\checkmark$	~	LR, DT, SVM, RF, KNN, RNN	RSSI, CIR, SINR, interference power, type of interference	
Channel access method	Aloha	Aloha	Aloha	Aloha	Logistic regression, SVM, DT, KNN, Naive Bayes, DNN, Ensemble	LBT, retransmission limit, DC, channel type, no. of EDs	
Confirmed mode (ACK)	×	×	x	✓	Logistic regression, SVM, DT, KNN, DNN, Ensemble	Retransmission count, ACK, PSR, DR, ToA, Link margin, history of packets	
Unconfirmed mode (no-ACK)	$\checkmark$	~	~	✓	SVM, DT, KNN, DNN, En- semble	DR, SNR, RSSI, Link margin, his- tory of packets, PER	

**Table 7.** Comparison of publicly available LoRaWAN frameworks for dataset collection and applicable ML methods based on the required features for resource classification.

X = not mentioned in the referenced paper;  $\checkmark$  = defined in the referenced paper.

LoRaSim is a valuable framework for understanding the behavior and performance of LoRaWAN. It can be used to design and optimize LoRaWAN networks to ensure reliable and efficient communication, as utilized in [29,30,70,141,173,181–196].

## 4.2. SimLoRaSF: Simulator for LoRa SF

SimLoRaSF is a LoRaWAN simulator that can be used to study the impact of different SFs on network performance [127,128]. It is a Python-based framework that uses a discreteevent simulation model. It works by creating a virtual LoRaWAN network and then simulating the transmission of packets between EDs in the network by keeping a track record of the packet transmission time, the SF, the transmitting source (e.g., ED), the packet size, the duration (e.g., ToA), and status of each transmission (e.g., ACK).

The packet transmission is categorized into three statuses: transmitted, interfered with, or under sensitivity. A packet is considered successful if it is received correctly by the GW. A packet transmission is considered interfered with if it is corrupted by interference from another transmission in the network. A packet transmission is considered under sensitivity if the SNR is too low for the destination node to receive it correctly.

# 4.3. LoRaWANSim: LoRaWAN Simulator

The LoRaWANSim [197,198] is a powerful simulator framework that can be used to study the behavior of LoRaWAN networks under PHY and MAC layer features. The framework provides a flexible simulation environment, allowing users to control PHY and MAC layer parameters. The PHY layer models the PHY transmission and reception of LoRa signals. It implements a complete LoRa transceiver, including the ability to generate modulated signals and perform demodulation tasks. It also considers factors such as interference and multiple GW scenarios. The MAC layer models the data traffic on the LoRaWAN network. It manages channel access, ensuring multiple EDs can share the same channel without interfering. Furthermore, it considers UL and downlink interference occurring over the same channel, DC limitations of 1%, and energy consumption.

## 4.4. ns-3: Network Simulator-3

The LoRaWAN ns-3 module [151,199] is among the most widely used simulators for utilizing ML, offering a wide range of LoRaWAN features, such as bi-directional communication, confirmed and unconfirmed modes, support for the ADR on the ED and NS-sides, buildings, DC limitations on the EU-868 MHz frequency, energy consumption module, etc. Furthermore, other promising features and modules, for instance, ADR in unconfirmed mode [7], mobility patterns [200], blind ADR [113] and ML methods [114,148] have been added by other researchers. However, they are not publicly available. Furthermore, this ns-3 module [151,199] has been extensively utilized in the literature [7,10,27,100,113,114,148,199–203].

## 4.5. Remarks

These dataset collection frameworks offer valuable tools for researchers and practitioners to analyze and optimize LoRaWAN networks. These frameworks enable the study of collision behavior [30], performance evaluation under different SFs, and their impact on the LoRaWAN network performance [127], investigation of PHY and MAC layer features [197], and a complete ns-3 framework for LoRaWAN analysis. The frameworks have been widely used in the literature, demonstrating their effectiveness in understanding and improving LoRaWAN network performance. Among these simulators, the state-of-the-art ns-3 module [151] has been utilized for intelligent SF allocation in the existing literature using DNN and multi-arm bandits approaches in [114,176].

# 5. LoRaWAN Datasets

The LoRa research community lacks a large dataset that can be used to study the resource allocation, interference, collision issues, and behavior of LoRaWAN networks in various conditions. The existing datasets are typically small and specific to a particular network or application owing to a few large-scale deployments of LoRaWAN. In addition, the data collected from LoRaWAN networks are not well-documented, making it challenging to use for research purposes. Despite these limitations, the existing datasets can be utilized to examine a variety of aspects of LoRaWAN networks. For example, the LoRaWAN dataset can be used to study the performance of LoRaWAN networks under different conditions, the impact of resource management on network performance, and the traffic patterns of LoRaWAN EDs. The rest of this section discusses the available datasets designed for different IoT applications.

#### 5.1. Localization

A large dataset for LoRaWAN and Sigfox was collected in urban and rural areas from 17 November 2017 to 5 February 2018, which contains the RSSI at each GW, latitude and longitude of the ED, and the SF utilized during data transmission [204]. The authors utilized the dataset in [205] for indoor localization, where the results showed that the mean location estimation error for Sigfox was recorded as 214.58 m and 688.97 m in rural and urban scenarios, respectively. In addition, the mean location estimation error for the urban LoRaWAN dataset was 398.40 m. The datasets presented in [204,205] can be leveraged to evaluate the performance of LoRaWAN-based indoor positioning systems by developing new fingerprinting algorithms. The dataset has been extensively utilized for LoRaWAN-based indoor positioning systems in [206–213]. Similarly, 100 RSSI values were collected for a target node and for the 11 anchors at LoS and NLoS in indoor and outdoor environments for improving the localization through LoRa measurements [214,215].

The authors in [216,217] generated a dataset called the "LoRaWAN at the Edge Dataset (LoED)." The LoED dataset [218] was collected under an urban scenario for four months, where nine GWs were utilized in central London. Among nine GWs, five were outdoor (line-of-sight (LoS)), four GWS located indoors with limited LoS, and one had no-LoS on the ground. The data was captured for a 2–4 month period generated by smart city applications. Overall, 11,263,001 packets were collected from 8503 unique EDs, comprised of the features

such as cyclic redundancy check (CRC), RSSI, SNR, SF, frequency, bandwidth, CR, packet type, device address, etc.

The LoRa RF fingerprinting dataset was collected considering indoor and outdoor scenarios comprising FFT and I/Q samples for indoor localization [219–221]. On the one hand, the dataset was collected for five consecutive days for indoor scenarios. During indoor dataset collection (e.g., room, outdoor, and office environments), 25 EDs were utilized, where every ED transmits 10 UL packets with an interval of 20 s. On the other hand, the outdoor dataset comprised four different scenarios regarding the distance: 5 m, 10 m, 15 m, and 20 m away from the GW.

The LoRaWAN performance mainly depends on several factors, including the distance between the ED and the GW, the underlying propagation environment, and the network traffic. To study the LoRaWAN network, the authors in [222] generated a dataset (publicly available, [223]) using 311 outdoor tests for 39 GW, which contains data regarding the longterm behavior of the LoRaWAN channel in Brno, Czech Republic. The dataset was collected under different environments, including urban, suburban, and rural areas, for two months, and included information about the SNR and RSSI concerning the closest GW distance.

The authors of [224] collected a fingerprinting dataset by conducting two studies on indoor and outdoor environments. The first study was conducted at the Brno University of Technology in Brno, Czech Republic, and the second was conducted at the University Politechnica of Bucharest in Bucharest, Romania. The dataset [225] comprises the RSSI information for various GWs using  $SF_7$ ,  $SF_9$ ,  $SF_{10}$ , and  $SF_{12}$ . The study showed reduced positioning accuracy in indoor and outdoor experiments. However, the SF dataset is limited since the authors have used only  $SF_7$ ,  $SF_9$ ,  $SF_{10}$ , and  $SF_{12}$ .

#### 5.2. Weather Forecast

Owing to the long range of LoRaWAN, it can often be utilized in outdoor environments. However, the performance of LoRaWAN is greatly affected by weather conditions, such as humidity, temperature, and atmospheric pressure. Therefore, the authors in [226] generated a dataset about the correlation between RSSI and SNR conditions from 8 LoRaWAN EDs and a weather station. The dataset [215] was collected for more than 80 days in a vineyard in Italy, including more than 190,000 records of RSSI, GPS coordinates, temperature, humidity, and pressure data. Researchers can use the dataset to develop new algorithms for studying and improving the performance of LoRaWAN under different weather conditions.

#### 5.3. Security Attacks

LoRaWAN networks can be vulnerable to security attacks. Therefore, the study in [227] investigated the use of LoRa metadata to detect the presence of security flaws within the network. The authors in [227] collected a dataset of LoRaWAN transmissions for two months with SNR and RSSI measurements. The data collected included normal, collided, and jammed metadata, where normal values are annotated as  $CRC_{OK}$  and DoS attacks were denoted as  $CRC_{BAD}$ . They utilized various ML methods (e.g., logistic regression, DT, RF, and XGBoost) to predict the presence of jamming and security attacks. Among the ML methods, XGBoost was the most accurate ML method for predicting security attacks, since it was found that SNR and RSSI can be used to pinpoint normal versus anomalous signals.

## 5.4. Signal Quality/Path Loss

When designing a LoRaWAN network, it is essential to consider the path loss between the EDs and the GWs. Several factors, including distance, frequency, and weather, can affect path loss. The authors in [228] presented a LoRaWAN measurement dataset collected in Medellin, Colombia. The dataset [229] was collected for about four months by considering only four EDs, which include information about path loss, the distance between ED and GW, frequency, temperature, relative humidity, barometric pressure, particulate matter, and energy consumption. Furthermore, the authors claimed that leveraging the dataset would enable the estimation of weather-induced variations in path loss for LoRaWAN deployments, leading to enhanced precision in tracking and positioning data and the development of more efficient energy reduction strategies.

The dataset in [230,231] was generated using two GW and six mobile EDs in a  $6 \times 6 \text{ km}^2$  urban area in Tsinghua University, Beijing, China. The dataset [232] was collected over four months comprising RSSI, PDR of the transmitted packets, locations of the ED and GW, and timestamps of each measurement.

The dataset [233,234] was collected in indoor and outdoor environments where, during the indoor data collection, the distance between the EDs and the GW varied from 5 to 50 m. The floor map illustrated the walls, doors, and windows between the EDs and the GW. In the outdoor environment, railway stations were used without considering obstacles between the EDs and the GW. The dataset includes features such as the timestamp, the SNR, and the RSSI of the received packet at the GW.

## 5.5. Smart City

The dataset [235,236] is divided into two parts: LoRa parameters and sensor readings. The LoRa parameters dataset contains the timestamp of packet transmission, the channel used in packet transmission, the device extended unique identifier (DevEUI), the SNR, the RSSI, and the frame counter (FCNT). In comparison, the sensor dataset includes data about the measured quantities, such as CO2, sound average, sound peak, motion, light, temperature, humidity, and battery levels. The Smart Campus dataset can be used for various applications, including time-series forecasting and number of people prediction.

#### 5.6. Resource Allocation

Mainly, the SF allocation is dependent on the underlying propagation environment, mobility, ED position, and sensitivity. Therefore, in our previous work [114,148], we generated a dataset using the state-of-the-art ns-3 module [151], which is publicly available [237]. The dataset was generated for 10 days with a UL period of 10 min, which mainly comprises ED locations (i.e., X and Y coordinates), RSSI, SNR, the distance between ED and GW, and the ACK status of every UL packet transmitted with each SF. The dataset in [237] can be utilized for resource management (e.g., SF) for static and mobile EDs.

## 5.7. Remarks

In addition to existing datasets, few datasets are publicly available with limited documentation. For example, LoRaWAN traffic analysis dataset [238], outdoor experiments conducted for LoRa RSSI [239,240], LoRaWAN dataset using SF [241], and LoRa time series dataset [242]. Furthermore, apart from [237] dataset, these datasets [205,215,218,221,223,225] are not designed for resource management. However, they can be tested for resource allocation since most datasets include RSSI, SNR, and device location information, which are efficient features for resource management. In conclusion, Table 8 provides additional information regarding the datasets discussed in this section.

Table 8. Summary of existing datasets utilized for different applications of LoRaWAN.

Year	Paper Ref.	Tool Used	Dataset Variables	Size of Dataset	Purpose	ML Method
2018	[205]	Testbed	SF, RSSI, ED positions [204]	123,5229	Localization	KNN
2019	[215]	Testbed	RSSI [214]	x	Localization	Least squares
2020	[218]	Testbed	CRC, RSSI, SNR, SF, frequency, band-	11,263,001	Smart city, capacity	X
			width, coding rate, packet type [216]		planning	
2021	[221]	Testbed	IQ/FFT [220]	×	Localization	CNN
2021	[223]	Testbed	SNR, RSSI, Frequency [222]	×	X	X
2022	[225]	Testbed	RSSI [224]	Outdoor = 16,054, in-	Localization	KNN, LR, DT, SVM
				door = 7752		

Year	Paper Ref.	Tool Used	Dataset Variables	Size of Dataset	Purpose	ML Method
2022	[215]	Testbed	RSSI, GPS coordinates, along with weather data [226]	190,000	Weather forecast	×
2022	[227]	Testbed	RSSI, SNR, CRC	×	Security analysis	Logistic regression, DT, RF, and XGBoost
2022	[229]	Testbed	Timestamp, ED id, energy consumption, SF, SNR, distance, frequency, and other sensory data [228]	930,753	Positioning, energy behavior of LoRa	Linear regression
2022	[232]	Testbed	Timestamp, RSSI, PDR, locations of ED and GW [230,231]	×	Tracking	×
2023	[234]	Testbed	Timestamp, SNR, RSSI [233]	×	Locomotion mode recognition	Zero-Shot learning
2023	[236]	Testbed	Timestamp, SNR, RSSI, frame counter, and other sensory data [235]	×	Smart city, people counting	KNN, LSTM.
2023	[237]	ns-3 [151]	RSSI, SNR, ED positions [114]	108,401	SF management	DNN, LSTM, GRU

Table 8. Cont.

*X* = not mentioned in the referenced paper.

#### 6. ns-3-Based ML Frameworks

The ns-3 frameworks, mainly designed for ML, can be utilized or integrated with the LoRaWAN ns-3 module [151,199] to enhance the performance of LoRaWAN with ML. These frameworks are discussed in the remainder of this section.

#### 6.1. ns-3-AI Framework

Currently, researchers are interested in applying ML techniques to wireless communication networks [115–117]. It is owing to most ML techniques heavily relying on open-source TensorFlow and PyTorch ML frameworks. These two frameworks are developed independently and are extremely hard to merge. Moreover, connecting these two frameworks with data interaction is more reasonable and convenient. Therefore, the ns3-AI framework was proposed in [243,244]. The ns3-AI framework provides an efficient workflow between ns-3 and Python-based modules, enabling seamless data transfer and interaction between the two modules. As an example, using the ns-3-AI framework: (a) LSTM has been utilized to predict the channel quality, and (b) RL method for controlling the congestion occurring in TCP [244].

#### 6.2. ns-3-gym Framework

The ns3-gym is an open-source RL framework, integrating OpenAI Gym and ns-3 [245]. The OpenAI Gym is a popular and open-source RL toolkit providing an interface for interacting with RL environments. The OpenAI Gym offers predefined environments with well-defined state and action spaces, making developing and comparing RL algorithms easier. It supports various RL methods, allowing researchers from academia to focus on developing new learning algorithms. The ns-3 Gym fills the gap between ns-3 and OpenAI Gym by creating an interface. This interface allows researchers to leverage the capabilities of ns-3 within the OpenAI Gym framework. As a result, such integration enables researchers to apply RL techniques to network scenarios and train RL agents to make intelligent decisions in complex networking environments.

For example, the authors in [246] present two use cases of cognitive radio (CR) transmitters to solve the issue of radio channel selection in the IEEE 802.11 WLAN with external interference [247]. In case 1, the transmitter senses the entire BW, while case 2 is related to data transmission, where the transmitter monitors its channel to avoid collisions by selecting a channel free of interference.

## 6.3. Open Neural Network Exchange (ONNX) Framework

ML could be computationally costly in LoRaWAN owing to its low computational power. As a result, energy consumption increases and thereby reducing the network lifetime. Therefore, DL models can be trained on one-time generated data (using ns-3 or testbed), and the pre-trained model (e.g., inference model) can be utilized in LoRaWAN ns-3 modules [199]. The pre-trained model can be configured using an Open Neural Network Exchange (ONNX) [248]. ONNX is an open-source Application Programming Interface (API) based on the C++ programming language for DL and ML techniques. To use ONNX with ns-3, first, an ONNX-supported pre-trained model using TensorFlow and PyTorch can be generated. Then, the pre-trained model can be imported into ns-3 with the help of the ONNX API. The ONNX API will provide raw data during simulation (similar input samples used during ML model training), which can be used for resource management (e.g., SF or TP). One example of ONNX implementation in ns-3 can be found in [249], where it has been utilized to simulate and model the behavior of an Open Radio Access Network [250].

## 6.4. ns-3-FL: Federated Learning Framework

The ns-3-FL [251,252] is a new framework for simulating FL in a realistic network environment. FL is an ML technique that allows multiple devices to train a shared model without sharing their data. This is useful for privacy- and delay-sensitive applications like healthcare and finance.

The ns-3-FL framework is built on two existing simulators: FLSim [253,254] and ns-3. It provides a realistic and flexible way to simulate FL training and inference in various network settings. FLSim [253] is responsible for data distribution between client–server architecture and FL, whereas ns-3 simulates the network. As an example of ns-3-FL working, (1) the FLSim [253] requests network simulation and selects the number of EDs for the FL training round, (2) the ns-3 performs the simulation for the selected EDs, (3) the ns-3 transmits the latency and throughput of each ED to the FLSim [253], and (4) the FLSim utilizes the received data from ns-3 during computing the convergence time of the global model (using FedAvg and FedAsync algorithms [255,256]) and average throughput for this specified training round, as illustrated in Figure 3.



Figure 3. Working procedure of the ns-3-FL framework [252].

Furthermore, the ns-3-FL comprises three main components: the learning model, the network model, and the power control.

## 6.4.1. Learning Model

FL is an ML paradigm that allows multiple clients to train a shared model without sharing their data. The training is achieved by iteratively sending updates to a global model, which is then aggregated by the server using *FedAvg* and *FedAsync* algorithms [255,256]. The ns-3-FL supports two types of FL algorithms: synchronous and asynchronous. In synchronous FL, all clients update the model simultaneously, while in asynchronous FL, clients can update the model at different times.

## 6.4.2. Network Model

The ns-3 network simulator was used to model the latency and throughput between clients and the server, allowing to study of network conditions and their impact on the performance of FL.

#### 6.4.3. Power Model

A power model calculates the energy consumption of FL training. This model considers the number of multiply–accumulate operations performed by the clients and the energy used to transmit the model to the server.

## 6.5. AI-ERA LoRaWAN Framework

Our previously published AI-Efficient Resource Allocation (AI-ERA) framework for SF classification [114,237], designed using AI and ns-3 modules, is a powerful framework, as illustrated in Figure 4. In the AI-ERA framework, the DNN model comprises five fully connected layers, trained using X and Y coordinates, SNR, and received power ( $P_{rx}$ ). After achieving the desired level of classification accuracy, the pre-trained model is deployed on the ED side, where (1) a similar input sequence (utilized during training) is used as input to the pre-trained model, (2) the pre-trained model processes the input and classifies a suitable SF based on the learned knowledge, and (3) a mobile or static ED adapts the classified SF and start transmitting data in UL direction, as shown in Figure 4.



**Figure 4.** Illustration of the AI-ERA classification framework for spreading factor: (1) inference model deployed on the ED side, where ED prepares input sequence for the model, (2) SF classification by the pre-trained model, and (3) SF adaptation by the ED for uplink packet transmission [114].

In addition, the AI-ERA framework [237] provides three major components: AI module, dataset, and data labeling.

## 6.5.1. AI Module

As illustrated in Figure 4, the AI-ERA module is comprised of a pre-trained DNN model, which is deployed on the ED side for efficient SF classification to EDs during simulation.

## 6.5.2. Dataset

During the dataset generation, the AI-ERA framework utilized a regular ADR, where EDs transmit a packet with  $SF_7 \sim SF_{12}$  at a regular interval of 10 min, as shown in Figure 5. Over a period of 10 days, a dataset was generated with a UL interval of 10 min. The dataset mainly consists of the X and Y coordinates of the ED locations, along with RSSI, SNR,

the distance between ED and GW, and the ACK status of each SF in every UL packet transmitted.



Figure 5. Regular ADR utilized for dataset collection in AI-ERA framework.

#### 6.5.3. Data Labeling

To train the DNN model, the SF was labeled based on the successful ACK reception by the ED. There could be multiple ACK responses for the same packet transmitted with  $SF_7 \sim SF_{12}$ . As a result, the lowest SF is chosen for labeling, and a sequence of six groups as input is fed to the DNN model.

# 6.6. LoRaWAN Bandit Framework

LoRaWAN bandit provides an RL framework that can be utilized to allocate the optimal SF to EDs by leveraging a MAB approach [176,257]. The LoRaWAN bandit framework learns the trade-off between energy consumption for each SF. The framework is designed to work in two phases: exploration and exploitation. During the exploration phase, the framework utilizes all SF combinations and learns how the energy consumption and DR is affected during simulation. In the exploitation phase, the framework selects the optimal SF based on the learned information.

The framework utilizes delayed feedback, where it does not expect immediate feedback. However, it receives feedback after a delay caused by several factors, such as the time it takes to reach the GW and the time it takes to process the packet. The framework is implemented in the ns-3 simulator using the LoRaWAN module [151].

## 6.7. Remarks

The ns-3-based ML frameworks can be utilized by researchers and practitioners to analyze, optimize, and improve the performance of the LoRaWAN networks by efficiently managing the resource parameters. However, these ML frameworks [244,246,248,252] are designed for a different purpose. In addition, the ML frameworks in [114,148,237] and [176,257] are specifically designed for resource management utilizing the state-of-the-art LoRaWAN module [151]. These ML frameworks [114,148,176,237,257] can be adapted to improve the performance of LoRaWAN through resource management.

# 7. Discussion

In the current literature, ML-, DL-, and RL-based solutions have been extensively studied to tackle the critical challenge of resource allocation in LoRaWAN networks, aiming to enhance the performance of the LoRaWAN network. However, existing approaches predominantly focus on resolving resource allocation at the ED level or on the NS, leading to certain limitations and opportunities for improvement.

# 7.1. ML Methods at the ED Side

In the literature, several ML methods have been utilized at the ED side [114,127,148], bringing several advantages. The EDs can make autonomous decisions locally regarding

the SF selection without constant communication with the NS, thereby reducing communication overhead. In addition, latency is particularly advantageous in scenarios with limited network bandwidth, unstable connections, and the availability of limited communication channels. Moreover, utilizing ML methods for SF classification can efficiently improve the convergence period under static and mobile IoT EDs, thus improving the packet success ratio [114].

However, utilizing ML methods for resource management on the ED side can be disadvantageous. For example, the primary concern is the computational load it places on resource-constrained EDs. Low-power IoT devices with limited processing capabilities may struggle to support ML models due to the significant computational power and memory requirements. Therefore, extra computational power can lead to excessive energy consumption. Moreover, training and maintaining the models on each ED can be challenging, particularly when EDs have different hardware and software configurations, making it difficult to ensure model consistency across the LoRa network.

#### 7.2. ML Methods at the NS Side

When ML methods are utilized on the NS side, it can bring several advantages. For instance, the NS generally has more computational resources, allowing more complex and accurate models to be trained and deployed for resource management. Leveraging a centralized dataset, potentially containing data from massive EDs, can lead to a more comprehensive and accurate ML model for resource classification. Furthermore, updating and maintaining the ML model can be efficiently performed from a centralized location, ensuring consistency across the network.

However, implementing ML methods on the NS can have a negative impact on the network performance. In terms of mobility, if an ED receives a downlink *LinkADRReq* MAC command from the NS that contains a new SF, and when the ED transmits a packet with the updated SF, it may not be delivered to the GW, resulting in packet loss. This is because the underlying propagation environment changes significantly when the ED is mobile [10,114]. In addition, communication between the EDs and the NS for resource allocation decisions may introduce latency in a large-scale LoRaWAN deployment. Relying on the NS for every resource allocation decision might lead to single points of failure and reduce network performance in terms of increased convergence period.

#### 7.3. Remarks

Resource allocation through ML-, DL-, and RL-based solutions has promising potential to enhance the LoRaWAN network performance. However, addressing the challenges associated with dataset quality and considering the trade-offs of implementing the ML method on the EDs and the NS side is essential for developing effective and efficient resource allocation mechanisms in LoRaWAN networks. The choice between the two approaches must be made carefully, depending on the specific requirements and constraints of the LoRaWAN network. Both options present unique advantages and disadvantages that should be thoroughly evaluated for optimal performance and scalability.

## 8. Future Recommendations

This section presents future recommendations for SF and TP allocation to EDs using ML to improve the overall performance of LoRaWAN.

## 8.1. Spreading Factor Classification

The allocation of SF is dependent on few parameters, such as GW sensitivity ( $S_{GW}$ ) [27] and ED sensitivity ( $S_{ED}$ ) [105] thresholds (illustrated in Table 9), the distance between the GW and ED [7], path loss [129,146], ToA [133], interference/collision [201], ED positions with SNR [147,148], retransmission [10], the ratio of UL and downlink ACK [105], and packet drop ratio [258]. However, to apply ML/DL for SF allocation to EDs, multiple parameters have been adopted in the current literature, for example, the received power

(i.e., RSSI), SNR, the distance between ED and GW, and ED position [114]. For mobile EDs, the ML methods should be deployed on the ED side since the propagation environment changes drastically. Therefore, the decision of efficient SF selection by ED will help to deliver a packet successfully, thereby improving the packet success ratio. However, deploying the ML on the ED side will increase the computational cost as low-power devices with limited processing capabilities may struggle to support ML models due to the significant computational power and memory requirements. On the other hand, it is recommended to utilize the ML method on the NS side for static EDs. In such a case, the propagation environment remains unchanged [10].

SF	GW Sensitivity (S <sub>g</sub> ) [dBm]	ED Sensitivity (S <sub>e</sub> ) [dBm]	SNR [dB]
12	-142.5	-137.0	-20
11	-140.0	-135.0	-17.5
10	-137.5	-133.0	-15
9	-135.0	-130.0	-12.5
8	-132.5	-127.0	-10
7	-130.0	-124.0	-7.5

Table 9. Sensitivity and required SNR of EDs and GW with 125-kHz mode [27,200].

## 8.2. Transmission Power Classification

In LoRaWAN, an ML model can be trained for efficient TP allocation to EDs in an intelligent manner. The input features for TP classification might include *RSSI*, link quality indicator (LQI), energy consumption, interference in the channel, and distance between the GW and ED. The TP level in LoRaWAN ranges from 2 dBm to 14 dBm and can be divided into five possible classes (2, 5, 8, 11, 14) dBm [152,155,167]. A packet successfully delivered with any SF and TP level can be labeled with a particular TP. For example, a DNN model can be trained to classify an appropriate TP level for the EDs based on the provided inputs.

#### 8.3. Multiclass Multioutput Classification

Few existing works have investigated SF and TP classification problems individually. However, both SF and TP classification can be achieved simultaneously, which can be regarded as a multiclass multioutput classification problem, referred to as multi-resource classification (MRC). In the case of SF classification, there are six possible classes (i.e.,  $SF_7$ ,  $SF_8$ ,  $SF_9$ ,  $SF_{10}$ ,  $SF_{11}$ , and  $SF_{12}$ ). Similarly, the TP, ranging from 2 dBm to 14 dBm, can be classified into five possible classes (2, 5, 8, 11, 14) dBm [152,155,167]. During dataset collection, a packet would be transmitted with a pair of SF and TP. Based on the successful ACK reception by the ED for the lowest SF and TP pair, the features such as RSSI, SNR, the distance between ED and GW, ACK status, and X and Y coordinates will be labeled. Based on the requirements of the application (e.g., mobile or static), the ML/DL method can be implemented either on the ED side (in the case of mobile application) or on the NS side (for static application). It is because the underlying propagation environment changes drastically in the case of mobility, thereby the SF and TP selection decision should be taken by ED rather than NS.

#### 9. Conclusions

LoRaWAN is a promising IoT protocol for long-range and ultra-low power consumption applications. However, a few challenges need to be addressed before LoRa and LoRaWAN can be widely deployed, such as LoRa parameters configuration, interference, and optimized ADR. One promising approach to overcoming these challenges and improving the performance of LoRaWAN is ML. ML can be used to develop robust, efficient, and intelligent ADRs responsible for resource parameter configurations (e.g., SF, TP, CR, etc.). Furthermore, the field of ML in LoRa and LoRaWAN is growing fast, with recent research and development focusing on various areas such as SF and TP classification, collision analysis, and interference mitigation.

In conclusion, this survey provides a detailed analysis of the state-of-the-art ML/DL/RL applications utilized for improving the performance of LoRaWAN. Further, the survey discusses the publicly available dataset collection frameworks and publicly available datasets. For the identified required features, the use of potential ML methods has been determined for improving the performance of LoRaWAN. Furthermore, the ns-3-based ML frameworks have been highlighted that can be integrated with the widely adopted LoRaWAN ns-3 module. Finally, a discussion on current ML research efforts is highlighted with features utilized for ML, DL, and RL, along with future recommendations that show how the LoRaWAN performance can be further improved using ML techniques.

**Author Contributions:** Conceptualization, A.F.; methodology, A.F.; software, A.F.; validation, A.F. and J.-Y.P.; investigation, A.F. and J.-Y.P.; resources, A.F. and J.-Y.P.; data curation, A.F. and J.-Y.P.; writing—original draft preparation, A.F.; writing—review and editing, A.F. and J.-Y.P.; visualization, A.F.; supervision, A.F. and J.-Y.P.; project administration, A.F. and J.-Y.P.; funding acquisition, A.F. and J.-Y.P. All authors have read and agreed to the published version of the manuscript.

Funding: This work was supported by the Research Fund from Chosun University, 2022.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

#### References

- 1. Ducrot, N.; Ray, D.; Saadani, A.; Hersent, O.; Pop, G.; Remond, G. Lora Device Developer Guide. 2016. Available online: https://developer.orange.com/od-uploads/LoRa-Device-Developer-Guide-Orange.pdf (accessed on 16 June 2023).
- Telagam, N.; Kandasamy, N.; Ajitha, D. Smart Healthcare Monitoring System Using LoRaWAN IoT and Machine Learning Methods. In *Practical Artificial Intelligence for Internet of Medical Things*; CRC Press: Boca Raton, FL, USA, 2023; pp. 85–104.
- Islam, K.Z.; Murray, D.; Diepeveen, D.; Jones, M.G.; Sohel, F. Machine learning-based LoRa localisation using multiple received signal features. *IET Wirel. Sens. Syst.* 2023, 1–18. [CrossRef]
- 4. Farhad, A.; Pyun, J.Y. Resource Management for Massive Internet of Things in IEEE 802.11 ah WLAN: Potentials, Current Solutions, and Open Challenges. *Sensors* 2022, 22, 9509. [CrossRef]
- 5. Mekki, K.; Bajic, E.; Chaxel, F.; Meyer, F. A comparative study of LPWAN technologies for large-scale IoT deployment. *ICT Express* **2019**, *5*, 1–7. [CrossRef]
- 6. Gomez, C.; Veras, J.C.; Vidal, R.; Casals, L.; Paradells, J. A sigfox energy consumption model. Sensors 2019, 19, 681. [CrossRef]
- Farhad, A.; Kim, D.H.; Kim, B.H.; Mohammed, A.F.Y.; Pyun, J.Y. Mobility-Aware Resource Assignment to IoT Applications in Long-Range Wide Area Networks. *IEEE Access* 2020, *8*, 186111–186124. [CrossRef]
- Singh, R.K.; Puluckul, P.P.; Berkvens, R.; Weyn, M. Energy consumption analysis of LPWAN technologies and lifetime estimation for IoT application. Sensors 2020, 20, 4794. [CrossRef]
- RP002-1.0.2 LoRaWAN<sup>®</sup> Regional Parameters. 2020. Available online: https://hz137b.p3cdn1.secureserver.net/wp-content/ uploads/2020/11/RP\_2-1.0.2.pdf?time=1672853176 (accessed on 9 January 2023).
- Farhad, A.; Kim, D.H.; Pyun, J.Y. R-ARM: Retransmission-Assisted Resource Management in LoRaWAN for the Internet of Things. *IEEE Internet Things J.* 2022, 9, 7347–7361. [CrossRef]
- 11. Farhad, A.; Pyun, J.Y. Terahertz Meets AI: The State of the Art. Sensors 2023, 23, 5034.
- 12. Farahsari, P.S.; Farahzadi, A.; Rezazadeh, J.; Bagheri, A. A survey on indoor positioning systems for iot-based applications. *IEEE Internet Things J.* **2022**, *9*, 7680–7699. [CrossRef]
- Torres-Sospedra, J.; Gaibor, D.P.Q.; Nurmi, J.; Koucheryavy, Y.; Lohan, E.S.; Huerta, J. Scalable and Efficient Clustering for Fingerprint-Based Positioning. *IEEE Internet Things J.* 2022, 10, 3484–3499. [CrossRef]
- 14. LoRa Simulator (LoRaSim). Available online: https://github.com/AlexSartori/LoRaSim (accessed on 3 May 2023).
- 15. Zorbas, D.; Caillouet, C.; Abdelfadeel Hassan, K.; Pesch, D. Optimal data collection time in LoRa networks—A time-slotted approach. *Sensors* **2021**, *21*, 1193. [CrossRef]
- 16. Beltramelli, L.; Mahmood, A.; Österberg, P.; Gidlund, M.; Ferrari, P.; Sisinni, E. Energy efficiency of slotted LoRaWAN communication with out-of-band synchronization. *IEEE Trans. Instrum. Meas.* **2021**, *70*, 1–11. [CrossRef]

- Loh, F.; Mehling, N.; Metzger, F.; Hoßfeld, T.; Hock, D. LoRaPlan: A Software to Evaluate Gateway Placement in LoRaWAN. In Proceedings of the 17th International Conference on Network and Service Management (CNSM), Izmir, Turkey, 25–29 October 2021; pp. 385–387. [CrossRef]
- 18. Casals, L.; Gomez, C.; Vidal, R. The SF12 Well in LoRaWAN: Problem and End-Device-Based Solutions. Sensors 2021, 21, 6478. [CrossRef]
- Zorbas, D.; Abdelfadeel, K.; Kotzanikolaou, P.; Pesch, D. TS-LoRa: Time-slotted LoRaWAN for the industrial Internet of Things. Comput. Commun. 2020, 153, 1–10. [CrossRef]
- 20. Abdelfadeel, K.Q.; Zorbas, D.; Cionca, V.; Pesch, D. *FREE*—Fine-grained scheduling for reliable and energy-efficient data collection in LoRaWAN. *IEEE Internet Things J.* **2019**, *7*, 669–683. [CrossRef]
- Ta, D.T.; Khawam, K.; Lahoud, S.; Adjih, C.; Martin, S. LoRa-MAB: A Flexible Simulator for Decentralized Learning Resource Allocation in IoT Networks. In Proceedings of the 2019 12th IFIP Wireless and Mobile Networking Conference (WMNC), Paris, France, 11–13 September 2019; pp. 55–62. [CrossRef]
- Reynders, B.; Wang, Q.; Pollin, S. A LoRaWAN module for ns-3: Implementation and evaluation. In Proceedings of the 10th Workshop on ns-3, Surathkal, India, 13–14 June 2018; pp. 61–68.
- To, T.H.; Duda, A. Simulation of LoRa in NS-3: Improving LoRa Performance with CSMA. In Proceedings of the IEEE International Conference on Communications (ICC), Kansas City, MO, USA, 20–24 May 2018; pp. 1–7. [CrossRef]
- Slabicki, M.; Premsankar, G.; Di Francesco, M. Adaptive configuration of lora networks for dense IoT deployments. In Proceedings of the NOMS 2018—2018 IEEE/IFIP Network Operations and Management Symposium, Taipei, Taiwan, 23–27 April 2018; pp. 1–9. [CrossRef]
- Bounceur, A.; Marc, O.; Lounis, M.; Soler, J.; Clavier, L.; Combeau, P.; Vauzelle, R.; Lagadec, L.; Euler, R.; Bezoui, M. CupCarbon-Lab: An IoT emulator. In Proceedings of the 15th IEEE Annual Consumer Communications & Networking Conference (CCNC), Las Vegas, NV, USA, 12–15 January 2018; pp. 1–2. [CrossRef]
- Croce, D.; Gucciardo, M.; Mangione, S.; Santaromita, G.; Tinnirello, I. Impact of LoRa imperfect orthogonality: Analysis of link-level performance. *IEEE Commun. Lett.* 2018, 22, 796–799. [CrossRef]
- Magrin, D.; Centenaro, M.; Vangelista, L. Performance evaluation of LoRa networks in a smart city scenario. In Proceedings of the 2017 IEEE International Conference on Communications (ICC), Paris, France, 21–25 May 2017; pp. 1–7. [CrossRef]
- 28. Van den Abeele, F.; Haxhibeqiri, J.; Moerman, I.; Hoebeke, J. Scalability analysis of large-scale LoRaWAN networks in ns-3. *IEEE Internet Things J.* 2017, 4, 2186–2198. [CrossRef]
- Pop, A.I.; Raza, U.; Kulkarni, P.; Sooriyabandara, M. Does Bidirectional Traffic Do More Harm Than Good in LoRaWAN Based LPWA Networks? In Proceedings of the GLOBECOM 2017—2017 IEEE Global Communications Conference, Singapore, 4–8 December 2017; pp. 1–6. [CrossRef]
- Bor, M.C.; Roedig, U.; Voigt, T.; Alonso, J.M. Do LoRa low-power wide-area networks scale? In Proceedings of the 19th ACM International Conference on Modeling, Analysis and Simulation of Wireless and Mobile Systems, Malta, 13–17 November 2016; pp. 59–67.
- 31. Sigfox Simulator. Available online: https://github.com/maartenweyn/lpwansimulation (accessed on 26 May 2023).
- 32. NS-3 Module for Sigfox. Available online: https://github.com/DEIS-Tools/ns3-sigfox (accessed on 26 May 2023).
- 33. Blockchain-Based IoT Simulator. Available online: https://github.com/sapgan/NS3-IoT-Simulator (accessed on 26 May 2023).
- 34. NS-3-Based NB-IoT Simulator Module. Available online: https://github.com/gdbranco/RA5G\_NS3 (accessed on 26 May 2023).
- NS-3-Based Framework for Narrow Band-IoT and LTE. Available online: https://github.com/a3794110/ns-3-NB-IoT (accessed on 30 May 2023).
- 36. NS-3-Based Framework for 5G. Available online: https://github.com/QiuYukang/5G-LENA (accessed on 26 May 2023).
- 37. LENA Framework for LTE and Vehicle-to-Everything (V2X). Available online: https://github.com/signetlabdei/lena-plus (accessed on 26 May 2023).
- Polese, M.; Centenaro, M.; Zanella, A.; Zorzi, M. M2M massive access in LTE: RACH performance evaluation in a Smart City scenario. In Proceedings of the 2016 IEEE International Conference on Communications (ICC), Kuala Lumpur, Malaysia, 22–27 May 2016; pp. 1–6. [CrossRef]
- 39. Sinha, R.S.; Wei, Y.; Hwang, S.H. A survey on LPWA technology: LoRa and NB-IoT. ICT Express 2017, 3, 14–21. [CrossRef]
- 40. Raza, U.; Kulkarni, P.; Sooriyabandara, M. Low power wide area networks: An overview. *IEEE Commun. Surv. Tutor.* 2017, 19, 855–873. [CrossRef]
- 41. Piroddi, A.; Torregiani, M. Machine Learning Applied to LoRaWAN Network for Improving Fingerprint Localization Accuracy in Dense Urban Areas. *Network* 2023, *3*, 199–217. [CrossRef]
- LoRaWAN L2 1.0.4 Specification. 2020. Available online: https://hz137b.p3cdn1.secureserver.net/wp-content/uploads/2021/1 1/LoRaWAN-Link-Layer-Specification-v1.0.4.pdf?time=1672853176 (accessed on 7 January 2023).
- Haxhibeqiri, J.; De Poorter, E.; Moerman, I.; Hoebeke, J. A survey of lorawan for iot: From technology to application. *Sensors* 2018, 18, 3995. [CrossRef]
- 44. Butun, I.; Pereira, N.; Gidlund, M. Security risk analysis of LoRaWAN and future directions. Future Internet 2018, 11, 3. [CrossRef]
- Saari, M.; bin Baharudin, A.M.; Sillberg, P.; Hyrynsalmi, S.; Yan, W. LoRa—A survey of recent research trends. In Proceedings of the 41st International Convention on Information and Communication Technology, Electronics and Microelectronics (MIPRO), Opatija, Croatia, 21–25 May 2018; pp. 872–877. [CrossRef]

- 46. Muteba, F.; Djouani, K.; Olwal, T. A comparative Survey Study on LPWA IoT Technologies: Design, considerations, challenges and solutions. *Procedia Comput. Sci.* 2019, 155, 636–641. [CrossRef]
- Khalifeh, A.; Aldahdouh, K.A.; Darabkh, K.A.; Al-Sit, W. A Survey of 5G Emerging Wireless Technologies Featuring LoRaWAN, Sigfox, NB-IoT and LTE-M. In Proceedings of the International Conference on Wireless Communications Signal Processing and Networking (WiSPNET), Chennai, India, 21–23 March 2019; pp. 561–566. [CrossRef]
- Sarker, V.K.; Queralta, J.P.; Gia, T.N.; Tenhunen, H.; Westerlund, T. A Survey on LoRa for IoT: Integrating Edge Computing. In Proceedings of the 2019 Fourth International Conference on Fog and Mobile Edge Computing (FMEC), Rome, Italy, 10–13 June 2019; pp. 295–300. [CrossRef]
- Ntseane, L.; Isong, B. Analysis of LoRa/LoRaWAN Challenges: Review. In Proceedings of the International Multidisciplinary Information Technology and Engineering Conference (IMITEC), Vanderbijlpark, South Africa, 21–22 November 2019; pp. 1–7. [CrossRef]
- 50. Ayoub, W.; Samhat, A.E.; Nouvel, F.; Mroue, M.; Prévotet, J.C. Internet of mobile things: Overview of lorawan, dash7, and nb-iot in lpwans standards and supported mobility. *IEEE Commun. Surv. Tutor.* **2018**, *21*, 1561–1581. [CrossRef]
- Qin, Z.; Li, F.Y.; Li, G.Y.; McCann, J.A.; Ni, Q. Low-Power Wide-Area Networks for Sustainable IoT. *IEEE Wirel. Commun.* 2019, 26, 140–145. [CrossRef]
- Marais, J.M.; Abu-Mahfouz, A.M.; Hancke, G.P. A Review of LoRaWAN Simulators: Design Requirements and Limitations. In Proceedings of the 2019 International Multidisciplinary Information Technology and Engineering Conference (IMITEC), Vanderbijlpark, South Africa, 21–22 November 2019; pp. 1–6. [CrossRef]
- 53. Ertürk, M.A.; Aydın, M.A.; Büyükakkaşlar, M.T.; Evirgen, H. A survey on LoRaWAN architecture, protocol and technologies. *Future Internet* **2019**, *11*, 216. [CrossRef]
- Gambiroža, J.Č.; Mastelić, T.; Šolić, P.; Čagalj, M. Capacity in LoRaWAN Networks: Challenges and Opportunities. In Proceedings of the 2019 4th International Conference on Smart and Sustainable Technologies (SpliTech), Split, Croatia, 18–21 June 2019; pp. 1–6. [CrossRef]
- Bouras, C.; Gkamas, A.; Katsampiris Salgado, S.A.; Kokkinos, V. Comparison of LoRa simulation environments. In Proceedings of the 14th International Conference on Broad-Band Wireless Computing, Communication and Applications (BWCCA-2019), Antwerp, Belgium, 7–9 November 2019; Springer: Berlin/Heidelberg, Germany, 2019; pp. 374–385.
- Kufakunesu, R.; Hancke, G.P.; Abu-Mahfouz, A.M. A Survey on Adaptive Data Rate Optimization in LoRaWAN: Recent Solutions and Major Challenges. Sensors 2020, 20, 5044. [CrossRef]
- Lehong, C.; Isong, B.; Lugayizi, F.; Abu-Mahfouz, A.M. A Survey of LoRaWAN Adaptive Data Rate Algorithms for Possible Optimization. In Proceedings of the 2nd International Multidisciplinary Information Technology and Engineering Conference (IMITEC), Kimberley, South Africa, 25–27 November 2020; pp. 1–9. [CrossRef]
- 58. Sundaram, J.P.S.; Du, W.; Zhao, Z. A survey on lora networking: Research problems, current solutions, and open issues. *IEEE Commun. Surv. Tutor.* 2019, 22, 371–388. [CrossRef]
- 59. Alenezi, M.; Chai, K.K.; Chen, Y.; Jimaa, S. Ultra-dense LoRaWAN: Reviews and challenges. IET Commun. 2020, 14, 1361–1371. [CrossRef]
- 60. Staikopoulos, A.; Kanakaris, V.; Papakostas, G.A. Image Transmission via LoRa Networks—A Survey. In Proceedings of the IEEE 5th International Conference on Image, Vision and Computing (ICIVC), Beijing, China, 10–12 July 2020; pp. 150–154. [CrossRef]
- Cotrim, J.R.; Kleinschmidt, J.H. LoRaWAN mesh networks: A review and classification of multihop communication. *Sensors* 2020, 20, 4273. [CrossRef]
- 62. Noura, H.; Hatoum, T.; Salman, O.; Yaacoub, J.P.; Chehab, A. LoRaWAN security survey: Issues, threats and possible mitigation techniques. *Internet Things* **2020**, *12*, 100303. [CrossRef]
- Osorio, A.; Calle, M.; Soto, J.D.; Candelo-Becerra, J.E. Routing in LoRaWAN: Overview and challenges. *IEEE Commun. Mag.* 2020, 58, 72–76. [CrossRef]
- 64. Marais, J.M.; Abu-Mahfouz, A.M.; Hancke, G.P. A Survey on the Viability of Confirmed Traffic in a LoRaWAN. *IEEE Access* 2020, *8*, 9296–9311. [CrossRef]
- 65. Raychowdhury, A.; Pramanik, A. Survey on LoRa Technology: For Internet of Things. In *Intelligent Systems, Technologies and Applications;* Thampi, S.M., Trajkovic, L., Mitra, S., Nagabhushan, P., El-Alfy, E.S.M., Bojkovic, Z., Mishra, D., Eds.; Springer: Berlin/Heidelberg, Germany, 2020; pp. 259–271.
- 66. Abd Elkarim, S.I.; Basem, M. Machine Learning Approaches for LoRa Networks: A Survey. Available online: https: //www.researchgate.net/profile/Basem-Elhalawany/publication/370510570\_Machine\_Learning\_Approaches\_for\_LoRa\_ Networks\_A\_survey/links/645b475ef3512f1cc58856de/Machine-Learning-Approaches-for-LoRa-Networks-A-survey.pdf (accessed on 19 July 2023).
- 67. Silva, F.S.D.; Neto, E.P.; Oliveira, H.; Rosário, D.; Cerqueira, E.; Both, C.; Zeadally, S.; Neto, A.V. A Survey on Long-Range Wide-Area Network Technology Optimizations. *IEEE Access* **2021**, *9*, 106079–106106. [CrossRef]
- Ghazali, M.H.M.; Teoh, K.; Rahiman, W. A systematic review of real-time deployments of UAV-based Lora communication network. *IEEE Access* 2021, 9, 124817–124830. [CrossRef]
- 69. Benkahla, N.; Tounsi, H.; Song, Y.Q.; Frikha, M. Review and experimental evaluation of ADR enhancements for LoRaWAN networks. *Telecommun. Syst.* 2021, 77, 1–22. [CrossRef]
- 70. Lalle, Y.; Fourati, M.; Fourati, L.C.; Barraca, J.P. Routing Strategies for LoRaWAN Multi-Hop Networks: A Survey and an SDN-Based Solution for Smart Water Grid. *IEEE Access* **2021**, *9*, 168624–168647. [CrossRef]
- Ugwuanyi, S.; Paul, G.; Irvine, J. Survey of IoT for developing countries: Performance analysis of LoRaWAN and cellular NB-IoT networks. *Electronics* 2021, 10, 2224. [CrossRef]

- 72. Gkotsiopoulos, P.; Zorbas, D.; Douligeris, C. Performance Determinants in LoRa Networks: A Literature Review. *IEEE Commun. Surv. Tutor.* 2021, 23, 1721–1758. [CrossRef]
- da Silva, J.C.; Flor, D.d.L.; de Sousa Junior, V.A.; Bezerra, N.S.; de Medeiros, A.A. A Survey of LoRaWAN Simulation Tools in ns-3. J. Commun. Inf. Syst. 2021, 36, 17–30. [CrossRef]
- 74. Maurya, P.; Singh, A.; Kherani, A.A. A review: Spreading factor allocation schemes for LoRaWAN. *Telecommun. Syst.* 2022, 80, 449–468. [CrossRef]
- 75. Almuhaya, M.A.; Jabbar, W.A.; Sulaiman, N.; Abdulmalek, S. A survey on Lorawan technology: Recent trends, opportunities, simulation tools and future directions. *Electronics* 2022, *11*, 164. [CrossRef]
- Li, C.; Cao, Z. Lora networking techniques for large-scale and long-term iot: A down-to-top survey. ACM Comput. Surv. (CSUR) 2022, 55, 1–36. [CrossRef]
- Banti, K.; Karampelia, I.; Dimakis, T.; Boulogeorgos, A.A.A.; Kyriakidis, T.; Louta, M. LoRaWAN Communication Protocols: A Comprehensive Survey under an Energy Efficiency Perspective. *Telecom* 2022, *3*, 322–357. [CrossRef]
- Cheikh, I.; Aouami, R.; Sabir, E.; Sadik, M.; Roy, S. Multi-Layered Energy Efficiency in LoRa-WAN Networks: A Tutorial. *IEEE Access* 2022, 10, 9198–9231. [CrossRef]
- Idris, S.; Karunathilake, T.; Förster, A. Survey and Comparative Study of LoRa-Enabled Simulators for Internet of Things and Wireless Sensor Networks. *Sensors* 2022, 22, 5546. [CrossRef] [PubMed]
- Almuhaya, M.A.; Jabbar, W.A.; Sulaiman, N.; Sulaiman, A. An Overview on LoRaWAN Technology Simulation Tools. In International Conference of Reliable Information and Communication Technology; Springer: Berlin/Heidelberg, Germany, 2022; Volume 127, pp. 345–358.
- Hessel, F.; Almon, L.; Hollick, M. LoRaWAN Security: An Evolvable Survey on Vulnerabilities, Attacks and their Systematic Mitigation. ACM Trans. Sens. Netw. 2022, 18, 1–55. [CrossRef]
- Sun, Z.; Yang, H.; Liu, K.; Yin, Z.; Li, Z.; Xu, W. Recent advances in lora: A comprehensive survey. ACM Trans. Sensor Netw. 2022, 18, 1–44. [CrossRef]
- Jouhari, M.; Saeed, N.; Alouini, M.S.; Amhoud, E.M. A Survey on Scalable LoRaWAN for Massive IoT: Recent Advances, Potentials, and Challenges. *IEEE Commun. Surv. Tutor.* 2023. [CrossRef]
- 84. Baker, S.; Xiang, W. Artificial Intelligence of Things for Smarter Healthcare: A Survey of Advancements, Challenges, and Opportunities. *IEEE Commun. Surv. Tutor.* **2023**, *25*, 1261–1293. [CrossRef]
- 85. Kitchenham, B.A.; Charters, S. *Guidelines for Performing Systematic Literature Reviews in Software Engineering*; Technical Report EBSE 2007-001, Keele University and Durham University Joint Report; Department of Computer Science, University of Durham: Durham, UK, 2007.
- Semtech. LoRa® and LoRaWAN®: A Technical Overview. Available online: https://lora-developers.semtech.com/uploads/ documents/files/LoRa\_and\_LoRaWAN-A\_Tech\_Overview-Downloadable.pdf (accessed on 16 June 2023).
- 87. Pasolini, G. On the LoRa chirp spread spectrum modulation. Signal properties and their impact on transmitter and receiver architectures. *IEEE Trans. Wirel. Commun.* 2021, 21, 357–369. [CrossRef]
- 88. Goursaud, C.; Gorce, J.M. Dedicated networks for IoT: PHY/MAC state of the art and challenges. *EAI Endorsed Trans. Internet Thingsl.* **2015**, *1*, 1–11.
- 89. Xie, B.; Yin, Y.; Xiong, J. Pushing the Limits of Long Range Wireless Sensing with LoRa. *Proc. ACM Interact. Mob. Wearable Ubiquitous Technol.* 2021, *5*, 1–21. [CrossRef]
- Avila-Campos, P.; Astudillo-Salinas, F.; Vazquez-Rodas, A.; Araujo, A. Evaluation of LoRaWAN transmission range for wireless sensor networks in riparian forests. In Proceedings of the 22nd International ACM Conference on Modeling, Analysis and Simulation of Wireless and Mobile Systems, Miami Beach, FL, USA, 25–29 November 2019; pp. 199–206.
- 91. Lavric, A.; Popa, V. Internet of things and LoRa<sup>™</sup> low-power wide-area networks: A survey. In Proceedings of the 2017 International Symposium on Signals, Circuits and Systems (ISSCS), Iasi, Romania, 13–14 July 2017; pp. 1–5.
- LoRaWAN Regional Parameters (RP002-1.0.3). 2021. Available online: https://hz137b.p3cdn1.secureserver.net/wp-content/ uploads/2021/05/RP002-1.0.3-FINAL-1.pdf?time=1672853176 (accessed on 8 January 2023).
- Marais, J.M.; Malekian, R.; Abu-Mahfouz, A.M. LoRa and LoRaWAN testbeds: A review. In Proceedings of the IEEE AFRICON, Cape Town, South Africa, 18–20 September 2017; pp. 1496–1501. [CrossRef]
- Cuomo, F.; Gámez, J.C.C.; Maurizio, A.; Scipione, L.; Campo, M.; Caponi, A.; Bianchi, G.; Rossini, G.; Pisani, P. Towards traffic-oriented spreading factor allocations in LoRaWAN systems. In Proceedings of the 2018 17th Annual Mediterranean Ad Hoc Networking Workshop (Med-Hoc-Net), Capri, Italy, 20–22 June 2018; pp. 1–8. [CrossRef]
- 95. Fair Access Policy. Available online: https://www.thethingsnetwork.org/forum/t/fair-use-policy-explained/1300 (accessed on 9 January 2023).
- Migabo, E.; Djouani, K.; Kurien, A.; Olwal, T. A comparative survey study on LPWA networks: LoRa and NB-IoT. In Proceedings of the Future Technologies Conference (FTC), Vancouver, BC, Canada, 29–30 November 2017; pp. 29–30.
- de Carvalho Silva, J.; Rodrigues, J.J.P.C.; Alberti, A.M.; Solic, P.; Aquino, A.L.L. LoRaWAN—A low power WAN protocol for Internet of Things: A review and opportunities. In Proceedings of the 2017 2nd International Multidisciplinary Conference on Computer and Energy Science (SpliTech), Split, Croatia, 12–14 July 2017; pp. 1–6.

- Farhad, A.; Kwon, G.R.; Pyun, J.Y. Mobility Adaptive Data Rate Based on Kalman Filter for LoRa-Empowered IoT Applications. In Proceedings of the 2023 IEEE 20th Consumer Communications & Networking Conference (CCNC), Las Vegas, NV, USA, 8–11 January 2023; pp. 321–324. [CrossRef]
- Semtech. LoRaWAN Mobile Applications: Blind ADR. 2019. Available online: https://lora-developers.semtech.com/ documentation/tech-papers-and-guides/blind-adr/ (accessed on 31 May 2023).
- Farhad, A.; Kim, D.H.; Kwon, D.; Pyun, J.Y. An Improved Adaptive Data Rate for LoRaWAN Networks. In Proceedings of the 2020 IEEE International Conference on Consumer Electronics—Asia (ICCE-Asia), Seoul, Republic of Korea, 1–3 November 2020; pp. 1–4. [CrossRef]
- 101. Airtime Calculator for LoRaWAN. Available online: https://avbentem.github.io/airtime-calculator/ttn/eu868/51,12 (accessed on 9 January 2023).
- 102. TTN Airtime Calculator for LoRaWAN. Available online: https://www.thethingsnetwork.org/airtime-calculator (accessed on 9 January 2023).
- Bor, M.; Roedig, U. LoRa Transmission Parameter Selection. In Proceedings of the 2017 13th International Conference on Distributed Computing in Sensor Systems (DCOSS), Ottawa, ON, Canada, 5–7 June 2017; pp. 27–34. 2017.10. [CrossRef]
- Al-Gumaei, Y.A.; Aslam, N.; Chen, X.; Raza, M.; Ansari, R.I. Optimizing Power Allocation in LoRaWAN IoT Applications. *IEEE Internet Things J.* 2022, *9*, 3429–3442. [CrossRef]
- 105. Farhad, A.; Kim, D.H.; Pyun, J.Y. Resource Allocation to Massive Internet of Things in LoRaWANs. Sensors 2020, 20, 2645. [CrossRef]
- 106. Marini, R.; Cerroni, W.; Buratti, C. A Novel Collision-Aware Adaptive Data Rate Algorithm for LoRaWAN Networks. IEEE Internet Things J. 2021, 8, 2670–2680. [CrossRef]
- 107. Anwar, K.; Rahman, T.; Zeb, A.; Khan, I.; Zareei, M.; Vargas-Rosales, C. RM-ADR: Resource Management Adaptive Data Rate for Mobile Application in LoRaWAN. Sensors 2021, 21, 7980. [CrossRef]
- Moysiadis, V.; Lagkas, T.; Argyriou, V.; Sarigiannidis, A.; Moscholios, I.D.; Sarigiannidis, P. Extending ADR mechanism for LoRa enabled mobile end-devices. *Simul. Model. Pract. Theory* 2021, *113*, 102388. [CrossRef]
- Park, J.; Park, K.; Bae, H.; Kim, C.K. EARN: Enhanced ADR with Coding Rate Adaptation in LoRaWAN. *IEEE Internet Things J.* 2020, 7, 11873–11883. [CrossRef]
- Benkahla, N.; Tounsi, H.; Ye-Qiong, S.; Frikha, M. Enhanced ADR for LoRaWAN networks with mobility. In Proceedings of the 2019 15th International Wireless Communications & Mobile Computing Conference (IWCMC), Tangier, Morocco, 24–28 June 2019; pp. 1–6.
- 111. Semtech. Understanding the LoRa Adaptive Data Rate. Available online: https://lora-developers.semtech.com/uploads/ documents/files/Understanding\_LoRa\_Adaptive\_Data\_Rate\_Downloadable.pdf (accessed on 16 June 2023).
- 112. ETSI. System Reference Document (SRdoc); Technical Characteristics for Low Power Wide Area Networks and Chirp Spread Spectrum (LPWAN-CSS) Operating in the UHF Spectrum below 1 GHz; ETSI TR 103 526 V1.1.1 (2018-04). Available online: https://www.etsi.org/deliver/etsi\_tr/103500\_103599/103526/01.01.01\_60/tr\_103526v010101p.pdf (accessed on 16 June 2023).
- 113. Farhad, A.; Kim, D.H.; Yoon, J.S.; Pyun, J.Y. Feasibility Study of the LoRaWAN blind Adaptive Data Rate. In Proceedings of the Twelfth International Conference on Ubiquitous and Future Networks (ICUFN), Jeju Island, Republic of Korea, 17–20 August 2021; pp. 67–69. [CrossRef]
- 114. Farhad, A.; Pyun, J.Y. AI-ERA: Artificial Intelligence-Empowered Resource Allocation for LoRa-Enabled IoT Applications. *IEEE Trans. Ind. Inform.* 2023, 1–13.
- Yu, P.; Zhou, F.; Zhang, X.; Qiu, X.; Kadoch, M.; Cheriet, M. Deep Learning-Based Resource Allocation for 5G Broadband TV Service. *IEEE Trans. Broadcast.* 2020, 66, 800–813. [CrossRef]
- Ye, H.; Li, G.Y.; Juang, B.H.F. Deep Reinforcement Learning Based Resource Allocation for V2V Communications. *IEEE Trans. Veh. Technol.* 2019, 68, 3163–3173. [CrossRef]
- Ahmed, K.I.; Tabassum, H.; Hossain, E. Deep Learning for Radio Resource Allocation in Multi-Cell Networks. *IEEE Netw.* 2019, 33, 188–195. [CrossRef]
- Movassagh, A.A.; Alzubi, J.A.; Gheisari, M.; Rahimi, M.; Mohan, S.; Abbasi, A.A.; Nabipour, N. Artificial neural networks training algorithm integrating invasive weed optimization with differential evolutionary model. *J. Ambient. Intell. Humaniz. Comput.* 2021, 4, 6017–6025. [CrossRef]
- 119. Otynshin, A. Performance Enhancements of LoRaWAN Using Machine Learning on the Edge. Ph.D. Thesis, Nazarbayev University, Astana, Kazakhstan, 2021.
- 120. Rodić, L.D.; Perković, T.; Škiljo, M.; Šolić, P. Privacy leakage of LoRaWAN smart parking occupancy sensors. *Future Gener. Comput. Syst.* **2023**, 138, 142–159. [CrossRef]
- 121. Perković, T.; Dujić Rodić, L.; Šabić, J.; Šolić, P. Machine Learning Approach towards LoRaWAN Indoor Localization. *Electronics* 2023, 12, 457. [CrossRef]
- 122. Simeone, O. A Very Brief Introduction to Machine Learning With Applications to Communication Systems. *IEEE Trans. Cogn. Commun. Netw.* **2018**, *4*, 648–664. [CrossRef]
- 123. Hussain, F.; Hassan, S.A.; Hussain, R.; Hossain, E. Machine Learning for Resource Management in Cellular and IoT Networks: Potentials, Current Solutions, and Open Challenges. *IEEE Commun. Surv. Tutor.* **2020**, 22, 1251–1275. [CrossRef]

- 124. Paredes, W.D.; Kaushal, H.; Vakilinia, I.; Prodanoff, Z. LoRa Technology in Flying Ad Hoc Networks: A Survey of Challenges and Open Issues. *Sensors* 2023, 23, 2403. [CrossRef]
- 125. Gomez, C.A.; Shami, A.; Wang, X. Machine learning aided scheme for load balancing in dense IoT networks. *Sensors* 2018, 18, 3779. [CrossRef]
- 126. TTN Mapper. 2020. Available online: https://ttnmapper.org/heatmap/ (accessed on 1 June 2023).
- 127. Yatagan, T.; Oktug, S. Smart spreading factor assignment for lorawans. In Proceedings of the 2019 IEEE Symposium on Computers and Communications (ISCC), Barcelona, Spain, 29 June–3 July 2019; pp. 1–7. [CrossRef]
- 128. Simulator for LoRa Spreading Factor (SimLoRaSF). Available online: https://github.com/tugrulyatagan/simlorasf (accessed on 26 May 2023).
- 129. Minhaj, S.U.; Mahmood, A.; Abedin, S.F.; Hassan, S.A.; Bhatti, M.T.; Ali, S.H.; Gidlund, M. Intelligent Resource Allocation in LoRaWAN Using Machine Learning Techniques. *IEEE Access* **2023**, *11*, 10092–10106. [CrossRef]
- 130. Voigt, T.; Bor, M.; Roedig, U.; Alonso, J. Mitigating Inter-Network Interference in LoRa Networks. *arXiv* 2016, arXiv:1611.00688. [CrossRef]
- 131. LoRa Simulator. Available online: https://mcbor.github.io/lorasim/ (accessed on 3 June 2023).
- González-Palacio, M.; Tobón-Vallejo, D.; Sepúlveda-Cano, L.M.; Rúa, S.; Le, L.B. Machine-Learning-Based Combined Path Loss and Shadowing Model in LoRaWAN for Energy Efficiency Enhancement. *IEEE Internet Things J.* 2023, 10, 10725–10739. [CrossRef]
- 133. Spathi, K.P.; Beletsioti, G.A.; Kantelis, K.F.; Valkanis, A.; Nicopolitidis, P.; Papadimitriou, G.I. Increasing device energy efficiency in LoRaWAN networks via a learning-automata-based approach. *Int. J. Sens. Netw.* **2023**, *42*, 87–101. [CrossRef]
- 134. Kim, D.Y.; Kim, S. Data transmission using K-means clustering in low power wide area networks with mobile edge cloud. *Wirel. Pers. Commun.* **2019**, *105*, 567–581. [CrossRef]
- 135. Cuomo, F.; Garlisi, D.; Martino, A.; Martino, A. Predicting LoRaWAN behavior: How machine learning can help. *Computers* **2020**, *9*, 60. [CrossRef]
- 136. Tsakmakis, A.; Valkanis, A.; Beletsioti, G.; Kantelis, K.; Nicopolitidis, P.; Papadimitriou, G. Learning-Automata-Based Hybrid Model for Event Detection in LoRaWAN Networks. In Proceedings of the 2022 IEEE 21st Mediterranean Electrotechnical Conference (MELECON), Palermo, Italy, 14–16 June 2022; pp. 1183–1188. [CrossRef]
- 137. Elbsir, H.; Kassab, M.; Bhiri, S.; Bedoui, M.H.; Castells-Rufas, D.; Carrabina, J. LoRaWAN Optimization using optimized Auto-Regressive algorithm, Support Vector Machine and Temporal Fusion Transformer for QoS ensuring. In Proceedings of the 2022 18th International Conference on Wireless and Mobile Computing, Networking and Communications (WiMob), Thessaloniki, Greece, 10–12 October 2022; pp. 302–307. [CrossRef]
- 138. Xiao, Y.; Song, Y.; Liu, J. Multi-Agent Deep Reinforcement Learning Based Resource Allocation for Ultra-Reliable Low-Latency Internet of Controllable Things. *IEEE Trans. Wirel. Commun.* 2023. [CrossRef]
- Chan, J.; Wang, A.; Krishnamurthy, A.; Gollakota, S. Deepsense: Enabling carrier sense in low-power wide area networks using deep learning. *arXiv* 2019, arXiv:1904.10607.
- 140. Xiao, Y.; Chen, Y.; Nie, M.; Zhu, T.; Liu, Z.; Liu, C. Exploring LoRa and Deep Learning-Based Wireless Activity Recognition. *Electronics* **2023**, *12*, 629. [CrossRef]
- 141. Cui, S.; Joe, I. Collision prediction for a low power wide area network using deep learning methods. *J. Commun. Netw.* 2020, 22, 205–214. [CrossRef]
- Alenezi, M.; Chai, K.K.; Alam, A.S.; Chen, Y.; Jimaa, S. Unsupervised learning clustering and dynamic transmission scheduling for efficient dense LoRaWAN networks. *IEEE Access* 2020, *8*, 191495–191509. [CrossRef]
- 143. Li, C.; Guo, H.; Tong, S.; Zeng, X.; Cao, Z.; Zhang, M.; Yan, Q.; Xiao, L.; Wang, J.; Liu, Y. NELoRa: Towards ultra-low SNR LoRa communication with neural-enhanced demodulation. In Proceedings of the 19th ACM Conference on Embedded Networked Sensor Systems, Coimbra, Portuga, 15–17 November 2021; pp. 56–68.
- 144. Shen, G.; Zhang, J.; Marshall, A.; Peng, L.; Wang, X. Radio frequency fingerprint identification for LoRa using spectrogram and CNN. In Proceedings of the IEEE INFOCOM 2021-IEEE Conference on Computer Communications, Vancouver, BC, Canada, 10–13 May 2021; pp. 1–10.
- 145. Al-Shawabka, A.; Pietraski, P.; Pattar, S.B.; Restuccia, F.; Melodia, T. DeepLoRa: Fingerprinting LoRa devices at scale through deep learning and data augmentation. In Proceedings of the Twenty-Second International Symposium on Theory, Algorithmic Foundations, and Protocol Design for Mobile Networks and Mobile Computing, Shanghai, China, 26–29 July 2021; ACM: New York, NY, USA, 2021; pp. 251–260.
- 146. Liu, L.; Yao, Y.; Cao, Z.; Zhang, M. DeepLoRa: Learning accurate path loss model for long distance links in LPWAN. In Proceedings of the IEEE INFOCOM 2021—IEEE Conference on Computer Communications, Vancouver, BC, Canada, 10–13 May 2021; pp. 251–260. [CrossRef]
- 147. Elkarim, S.I.A.; Elsherbini, M.; Mohammed, O.; Khan, W.U.; Waqar, O.; ElHalawany, B.M. Deep Learning Based Joint Collision Detection and Spreading Factor Allocation in LoRaWAN. In Proceedings of the 2022 IEEE 42nd International Conference on Distributed Computing Systems Workshops (ICDCSW), Bologna, Italy, 10 July 2022; pp. 187–192. [CrossRef]
- Farhad, A.; Kim, D.H.; Yoon, J.S.; Pyun, J.Y. Deep Learning-Based Channel Adaptive Resource Allocation in LoRaWAN. In Proceedings of the 2022 International Conference on Electronics, Information, and Communication (ICEIC), Jeju, Republic of Korea, 6–9 February 2022; pp. 1–5. [CrossRef]

- Lee, S.; Lee, J.; Hwang, J.; Choi, J.K. A Novel Deep Learning-Based IoT Device Transmission Interval Management Scheme for Enhanced Scalability in LoRa Networks. *IEEE Wirel. Commun. Lett.* 2021, 10, 2538–2542. [CrossRef]
- 150. Intel Berkeley Research Lab Data. Available online: http://db.csail.mit.edu/labdata/labdata.html (accessed on 19 July 2023).
- 151. LoRaWAN ns-3 Module. Available online: https://github.com/signetlabdei/lorawan (accessed on 30 May 2023).
- Fedullo, T.; Morato, A.; Tramarin, F.; Bellagente, P.; Ferrari, P.; Sisinni, E. Adaptive LoRaWAN transmission exploiting reinforcement learning: The industrial case. In Proceedings of the 2021 IEEE International Workshop on Metrology for Industry 4.0 & IoT (MetroInd4. 0&IoT), Rome, Italy, 7–9 June 2021; pp. 671–676.
- 153. Hamdi, R.; Baccour, E.; Erbad, A.; Qaraqe, M.; Hamdi, M. LoRa-RL: Deep reinforcement learning for resource management in hybrid energy LoRa wireless networks. *IEEE Internet Things J.* 2021, *9*, 6458–6476. [CrossRef]
- Carvalho, R.; Al-Tam, F.; Correia, N. Q-Learning ADR Agent for LoRaWAN Optimization. In Proceedings of the 2021 IEEE International Conference on Industry 4.0, Artificial Intelligence, and Communications Technology (IAICT), Bandung, Indonesia, 27–28 July 2021; pp. 104–108. [CrossRef]
- 155. Tellache, A.; Mekrache, A.; Bradai, A.; Boussaha, R.; Pousset, Y. Deep Reinforcement Learning based Resource Allocation in Dense Sliced LoRaWAN Networks. In Proceedings of the 2022 IEEE International Conference on Consumer Electronics (ICCE), Las Vegas, NV, USA, 7–9 January 2022; pp. 1–6. [CrossRef]
- 156. Soori, M.; Arezoo, B.; Dastres, R. Artificial intelligence, machine learning and deep learning in advanced robotics, a review. *Cogn. Robot.* **2023**, *3*, 54–70. [CrossRef]
- 157. Na, S.; Rouček, T.; Ulrich, J.; Pikman, J.; Krajník, T.s.; Lennox, B.; Arvin, F. Federated Reinforcement Learning for Collective Navigation of Robotic Swarms. *IEEE Trans. Cogn. Dev. Syst.* 2023. [CrossRef]
- Hellström, H.; da Silva, J.M.B., Jr.; Amiri, M.M.; Chen, M.; Fodor, V.; Poor, H.V.; Fischione, C. Wireless for machine learning: A survey. *Found. Trends*® Signal Process. 2022, 15, 290–399. [CrossRef]
- 159. Nisioti, E. Reinforcement Learning-Based Optimization of Multiple Access in Wireless Networks. Ph.D. Thesis, University of Essex, Colchester, UK , 2021.
- 160. Bonnefoi, R.; Moy, C.; Palicot, J. Improvement of the LPWAN AMI backhaul's latency thanks to reinforcement learning algorithms. *EURASIP J. Wirel. Commun. Netw.* 2018, 2018, 1–18. [CrossRef]
- Aihara, N.; Adachi, K.; Takyu, O.; Ohta, M.; Fujii, T. Reinforcement Learning Aided Orthogonal Frequency Allocation in LoRaWAN. *IEICE Tech. Rep.* 2019, 119, 45–46.
- Sandoval, R.M.; Garcia-Sanchez, A.J.; Garcia-Haro, J. Optimizing and updating lora communication parameters: A machine learning approach. *IEEE Trans. Netw. Serv. Manag.* 2019, 16, 884–895. [CrossRef]
- 163. Aihara, N.; Adachi, K.; Takyu, O.; Ohta, M.; Fujii, T. Q-learning aided resource allocation and environment recognition in LoRaWAN with CSMA/CA. *IEEE Access* 2019, *7*, 152126–152137. [CrossRef]
- 164. Simpy. Event Discrete Simulation for Python. Available online: https://simpy.readthedocs.io/en/latest/ (accessed on 26 May 2023).
- Bonnefoi, R.; Besson, L.; Manco-Vasquez, J.; Moy, C. Upper-Confidence Bound for Channel Selection in LPWA Networks with Retransmissions. In Proceedings of the 2019 IEEE Wireless Communications and Networking Conference Workshop (WCNCW), Marrakech, Morocco, 15–18 April 2019; pp. 1–7. [CrossRef]
- Ilahi, I.; Usama, M.; Farooq, M.O.; Janjua, M.U.; Qadir, J. LoRaDRL: Deep Reinforcement Learning Based Adaptive PHY Layer Transmission Parameters Selection for LoRaWAN. In Proceedings of the 2020 IEEE 45th Conference on Local Computer Networks (LCN), Sydney, NSW, Australia, 16–19 November 2020; pp. 457–460. [CrossRef]
- 167. Yu, Y.; Mroueh, L.; Li, S.; Terré, M. Multi-Agent Q-Learning Algorithm for Dynamic Power and Rate Allocation in LoRa Networks. In Proceedings of the 2020 IEEE 31st Annual International Symposium on Personal, Indoor and Mobile Radio Communications, London, UK, 31 August–3 September 2020; pp. 1–5. [CrossRef]
- Ilahi, I.; Usama, M.; Farooq, M.O.; Janjua, M.U.; Qadir, J. Intelligent resource allocation in dense lora networks using deep reinforcement learning. arXiv 2020, arXiv:2012.11867.
- Khalifeh, A.F.; Aldahdouh, K.; Alouneh, S. LoRaWAN energy optimization with security consideration. *Int. Arab J. Inf. Technol.* 2021, 18, 476–483. [CrossRef]
- Ta, D.T.; Khawam, K.; Lahoud, S.; Adjih, C.; Martin, S. LoRa-MAB: Toward an intelligent resource allocation approach for LoRaWAN. In Proceedings of the 2019 IEEE global communications conference (GLOBECOM), Waikoloa, HI, USA, 9– 13 December 2019; pp. 1–6.
- Callebaut, G.; Ottoy, G.; van der Perre, L. Cross-Layer Framework and Optimization for Efficient Use of the Energy Budget of IoT Nodes. In Proceedings of the IEEE Wireless Communications and Networking Conference (WCNC), Marrakesh, Morocco, 15–18 April 2019; pp. 1–6. [CrossRef]
- 172. Azizi, F.; Teymuri, B.; Aslani, R.; Rasti, M.; Tolvaneny, J.; Nardelli, P.H.J. MIX-MAB: Reinforcement Learning-based Resource Allocation Algorithm for LoRaWAN. In Proceedings of the 2022 IEEE 95th Vehicular Technology Conference: (VTC2022-Spring), Helsinki, Finland, 19–22 June 2022; pp. 1–6. [CrossRef]
- Zhong, H.; Ning, L.; Wang, J.; Suo, S.; Chen, L. Optimization of LoRa SF Allocation Based on Deep Reinforcement Learning. Wirel. Commun. Mob. Comput. 2022, 2022. [CrossRef]
- 174. Chen, M.; Mokdad, L.; Ben-Othman, J.; Fourneau, J.M. Dynamic Parameter Allocation With Reinforcement Learning for LoRaWAN. IEEE Internet Things J. 2023, 10, 10250–10265. [CrossRef]

- 175. Chen, M.; Mokdad, L.; Othman, J.B.; Fourneau, J.M. MULANE—A Lightweight Extendable Agent-oriented LoRaWAN Simulator with GUI. In Proceedings of the 2021 IEEE Symposium on Computers and Communications (ISCC), Athens, Greece, 5–8 September 2021, pp. 1–6. [CrossRef]
- Navas, R.E.; Dandachi, G.; Hadjadj-Aoul, Y.; Maillé, P. Energy-Aware Spreading Factor Selection in LoRaWAN Using Delayed-Feedback Bandits. In Proceedings of the International Federation for Information Processing (IFIP) Networking 2023 Conference (NETWORKING 2023), Barcelona, Spain, 12–15 June 2023.
- Zhao, G.; Lin, K.; Chapman, D.; Metje, N.; Hao, T. Optimizing energy efficiency of LoRaWAN-based wireless underground sensor networks: A multi-agent reinforcement learning approach. *Internet Things* 2023, 22, 100776. [CrossRef]
- 178. Garrido-Hidalgo, C.; Fürst, J.; Roda-Sanchez, L.; Olivares, T.; Fernández-Caballero, A. Lessons Learned on the Design of a Predictive Agent for LoRaWAN Network Planning. In *Proceedings of the International Conference on Practical Applications of Agents* and Multi-Agent Systems, The PAAMS Collection; PAAMS 2023; Lecture Notes in Computer Science; Springer: Berlin/Heidelberg, Germany, 2023; Volume 13955, pp. 88–99. \_8. [CrossRef]
- 179. Teymuri, B.; Serati, R.; Anagnostopoulos, N.A.; Rasti, M. LP-MAB: Improving the Energy Efficiency of LoRaWAN Using a Reinforcement-Learning-Based Adaptive Configuration Algorithm. *Sensors* **2023**, *23*, 2363. [CrossRef]
- Ayoub Kamal, M.; Alam, M.M.; Sajak, A.A.B.; Mohd Su'ud, M. Requirements, Deployments, and Challenges of LoRa Technology: A Survey. *Comput. Intell. Neurosci.* 2023, 2023. [CrossRef]
- Farooq, M.O.; Pesch, D. Evaluation of Multi-Gateway LoRaWAN with Different Data Traffic Models. In Proceedings of the IEEE 43rd Conference on Local Computer Networks (LCN), Chicago, IL, USA, 1–4 October 2018; pp. 279–282. [CrossRef]
- Francisco, S.; Pinho, P.; Luís, M. Improving LoRa Network Simulator for a More Realistic Approach on LoRaWAN. In Proceedings of the 2021 Telecoms Conference (ConfTELE), Leiria, Portugal, 11–12 February 2021; pp. 1–6. 50222.2021.9435570. [CrossRef]
- Sugianto, S.; Anhar, A.A.; Harwahyu, R.; Sari, R.F. Simulation of Mobile LoRa Gateway for Smart Electricity Meter. In Proceedings of the 2018 5th International Conference on Electrical Engineering, Computer Science and Informatics (EECSI), Malang, Indonesia, 16–18 October 2018; pp. 292–297. [CrossRef]
- 184. Spinsante, S.; Gioacchini, L.; Scalise, L. A novel experimental-based tool for the design of LoRa networks. In Proceedings of the 2019 II Workshop on Metrology for Industry 4.0 and IoT (MetroInd4.0 & IoT), Naples, Italy, 4–6 June 2019; pp. 317–322. [CrossRef]
- Borkotoky, S.; Bettstetter, C.; Schilcher, U.; Raffelsberger, C. Allocation of Repetition Redundancy in LoRa. In Proceedings of the European Wireless 2019—25th European Wireless Conference, Aarhus, Denmark, 2–4 May 2019; pp. 1–6.
- 186. Ferreira, C.M.S.; Oliveira, R.A.R.; Silva, J.S. Low-Energy Smart Cities Network with LoRa and Bluetooth. In Proceedings of the 2019 7th IEEE International Conference on Mobile Cloud Computing, Services, and Engineering (MobileCloud), Newark, CA, USA, 4–9 April 2019; pp. 24–29. [CrossRef]
- Lee, G.; Youn, J. Group-based Transmission Scheduling Scheme for Building LoRa-based Massive IoT. In Proceedings of the 2020 International Conference on Artificial Intelligence in Information and Communication (ICAIIC), Fukuoka, Japan, 19–21 February 2020; pp. 583–586. [CrossRef]
- Fam, P.A.; Faye, I. Towards More Energy Efficient MAC protocols for LoRaWAN Networks. In Proceedings of the 2020 XXXIIIrd General Assembly and Scientific Symposium of the International Union of Radio Science, Rome, Italy, 29 August–5 September 2020; pp. 1–4. [CrossRef]
- Sallum, E.; Pereira, N.; Alves, M.; Santos, M. Performance optimization on LoRa networks through assigning radio parameters. In Proceedings of the 2020 IEEE International Conference on Industrial Technology (ICIT), Buenos Aires, Argentina, 26–28 February 2020; pp. 304–309. [CrossRef]
- Charles, L.; Isong, B.; Lugayizi, F.; Abu-Mahfouz, A.M. Empirical Analysis of LoRaWAN-based Adaptive Data Rate Algorithms. In Proceedings of the IECON 2021—47th Annual Conference of the IEEE Industrial Electronics Society, Toronto, ON, Canada, 13–16 October 2021; pp. 1–7. [CrossRef]
- Wongwatthanaroek, K.; Silapunt, R. Transmission Sequencing to Improve LoRaWAN Performance. In Proceedings of the 2021 18th International Joint Conference on Computer Science and Software Engineering (JCSSE), Lampang, Thailand, 30 June–2 July 2021; pp. 1–5. [CrossRef]
- Alouneh, S.; Khalifeh, A.; Abou-Tair, D.E.D.I.; Aldahdouh, K.; Al-Hawari, F. An Open Source LoRaWAN Simulator Framework for the Internet of Things Applications. In Proceedings of the 2021 8th International Conference on Internet of Things: Systems, Management and Security (IOTSMS), Gandia, Spain, 6–9 December 2021; pp. 1–5. [CrossRef]
- Helou, G.; Ibrahim, M.; Tawil, R.; Mohanna, Y. Are Existing Analytical Models for LoRa Networks Accurate? In Proceedings of the 2022 4th IEEE Middle East and North Africa COMMunications Conference (MENACOMM), Amman, Jordan, 6–8 December 2022; pp. 24–31. [CrossRef]
- 194. Faye, I.; Fam, P.A.; Ndiaye, M.L. Energy consumption of IoT devices: An accurate evaluation to better predict battery lifetime. *Radio Sci.* 2022, *57*, 1–10. [CrossRef]
- 195. Lehong, C.; Isong, B.; Lugayizi, F.; Abu-Mahfouz, A. A spreading factor congestion status-aware adaptive data rate algorithm. *J. Sens. Actuator Netw.* **2021**, *10*, 70. [CrossRef]
- 196. Sallum, E.; Pereira, N.; Alves, M.; Santos, M. Improving Quality-of-Service in LoRa Low-Power Wide-Area Networks through Optimized Radio Resource Management. *J. Sens. Actuator Netw.* **2020**, *9*, 10. [CrossRef]

- 197. Marini, R.; Mikhaylov, K.; Pasolini, G.; Buratti, C. LoRaWanSim: A flexible simulator for LoRaWAN networks. *Sensors* 2021, 21, 695. [CrossRef] [PubMed]
- 198. LoRaWANSim: LoRaWAN Simulator. Available online: https://github.com/kvmikhayl/LoRaWAN\_simulator (accessed on 1 May 2023).
- 199. Magrin, D.; Capuzzo, M.; Zanella, A. A thorough study of LoRaWAN performance under different parameter settings. *IEEE Internet Things J.* **2019**, *7*, 116–127. [CrossRef]
- Farhad, A.; Kim, D.H.; Subedi, S.; Pyun, J.Y. Enhanced LoRaWAN Adaptive Data Rate for Mobile Internet of Things Devices. Sensors 2020, 20, 6466. [CrossRef] [PubMed]
- 201. Farhad, A.; Kim, D.H.; Sthapit, P.; Pyun, J.Y. Interference-Aware Spreading Factor Assignment Scheme for the Massive LoRaWAN Network. In Proceedings of the International Conference on Electronics, Information, and Communication (ICEIC), Auckland, New Zealand, 6 May 2019; pp. 1–2.
- Farhad, A.; Kim, D.; Pyun, J. Scalability of LoRaWAN in an Urban Environment: A Simulation Study. In Proceedings of the Eleventh International Conference on Ubiquitous and Future Networks (ICUFN), Zagreb, Croatia, 2–5 July 2019; pp. 677–681. [CrossRef]
- 203. Farhad, A.; Pyun, J.Y. HADR: A Hybrid Adaptive Data Rate in LoRaWAN for Internet of Things. ICT Express 2022, 8, 283–289. [CrossRef]
- Aernouts, M.; Berkvens, R.; Van Vlaenderen, K.; Weyn, M. Sigfox and LoRaWAN datasets for fingerprint localization in large urban and rural areas. *Data* 2018, 3, 13. [CrossRef]
- 205. Sigfox and LoRaWAN Datasets. Available online: https://zenodo.org/record/3342253 (accessed on 10 July 2023).
- Janssen, T.; Berkvens, R.; Weyn, M. Comparing Machine Learning Algorithms for RSS-Based Localization in LPWAN. In Advances on P2P, Parallel, Grid, Cloud and Internet Computing; Barolli, L., Hellinckx, P., Natwichai, J., Eds.; Springer: Berlin/Heidelberg, Germany, 2020; pp. 726–735.
- 207. Anagnostopoulos, G.G.; Kalousis, A. A Reproducible Comparison of RSSI Fingerprinting Localization Methods Using LoRaWAN. In Proceedings of the 2019 16th Workshop on Positioning, Navigation and Communications (WPNC), Bremen, Germany, 23–24 October 2019; pp. 1–6. [CrossRef]
- 208. Aernouts, M.; BniLam, N.; Berkvens, R.; Weyn, M. TDAoA: A combination of TDoA and AoA localization with LoRaWAN. *Internet Things* **2020**, *11*, 100236. [CrossRef]
- Li, Y.; Barthelemy, J.; Sun, S.; Perez, P.; Moran, B. Urban Vehicle Localization in Public LoRaWan Network. *IEEE Internet Things J.* 2022, 9, 10283–10294. [CrossRef]
- 210. Janssen, T.; Berkvens, R.; Weyn, M. Benchmarking RSS-based localization algorithms with LoRaWAN. *Internet Things* 2020, 11, 100235. [CrossRef]
- 211. Aqeel, I.; Iorkyase, E.; Zangoti, H.; Tachtatzis, C.; Atkinson, R.; Andonovic, I. LoRaWAN-implemented node localisation based on received signal strength indicator. *IET Wirel. Sens. Syst.* 2022. [CrossRef]
- Purohit, J.; Wang, X.; Mao, S.; Sun, X.; Yang, C. Fingerprinting-based Indoor and Outdoor Localization with LoRa and Deep Learning. In Proceedings of the GLOBECOM 2020—2020 IEEE Global Communications Conference, Taipei, Taiwan, 7–11 December 2020; pp. 1–6. [CrossRef]
- Liu, Y.; Tsang, K.F.; Zhu, H.; Chi, H.R.; Wei, Y.; Wang, H.; Wu, C.K. Efficient load balancing for heterogeneous radio-replicationcombined LoRaWAN. *IEEE Trans. Ind. Inform.* 2022, 18, 7400–7411. [CrossRef]
- Goldoni, E.; Prando, L.; Vizziello, A.; Savazzi, P.; Gamba, P. Experimental data set analysis of RSSI-based indoor and outdoor localization in LoRa networks. *Internet Technol. Lett.* 2019, 2, e75. [CrossRef]
- 215. Experimental Datasets of LoRa/LoRaWAN RSSI Measurements. Available online: https://github.com/emanueleg/lora-rssi (accessed on 11 July 2023).
- Bhatia, L.; Breza, M.; Marfievici, R.; McCann, J.A. LoED: The LoRaWAN at the edge dataset: Dataset. In Proceedings of the Third Workshop on Data: Acquisition To Analysis (ACM, New York, United States), Virtual Event, Japan, 16–19 November 2020; pp. 7–8. [CrossRef]
- Spadaccino, P.; Crinó, F.G.; Cuomo, F. LoRaWAN Behaviour Analysis through Dataset Traffic Investigation. Sensors 2022, 22, 2470. [CrossRef] [PubMed]
- 218. LoED: The LoRaWAN at the Edge Dataset: Dataset. Available online: https://zenodo.org/record/4121430 (accessed on 10 July 2023).
- 219. Elmaghbub, A.; Hamdaoui, B. LoRa device fingerprinting in the wild: Disclosing RF data-driven fingerprint sensitivity to deployment variability. *IEEE Access* 2021, 9, 142893–142909. [CrossRef]
- Elmaghbub, A.; Hamdaoui, B. Comprehensive RF Dataset Collection and Release: A Deep Learning-Based Device Fingerprinting Use Case. In Proceedings of the 2021 IEEE Globecom Workshops (GC Wkshps), Madrid, Spain, 7–11 December 2021; pp. 1–7. [CrossRef]
- 221. Release Note: Comprehensive LoRa RF Datasets for Device Fingerprinting Using Deep Learning. Available online: https://research.engr.oregonstate.edu/hamdaoui/sites/research.engr.oregonstate.edu.hamdaoui/files/release\_note\_rf\_ dataset\_oct2022.pdf (accessed on 11 July 2023).
- Masek, P.; Stusek, M.; Svertoka, E.; Pospisil, J.; Burget, R.; Lohan, E.S.; Marghescu, I.; Hosek, J.; Ometov, A. Measurements of LoRaWAN technology in urban scenarios: A data descriptor. *Data* 2021, 6, 62. [CrossRef]
- Measurements of LoRaWAN Technology in Urban Scenarios. Available online: https://github.com/BUTResearch/MDPI\_Data\_ Urban\_LPWA\_Measurement (accessed on 11 July 2023).
- Svertoka, E.; Rusu-Casandra, A.; Burget, R.; Marghescu, I.; Hosek, J.; Ometov, A. LoRaWAN: Lost for Localization? *IEEE Sens. J.* 2022, 22, 23307–23319. [CrossRef]

- 225. LoRaWAN: Lost for Localization? Available online: https://zenodo.org/record/7236698 (accessed on 11 July 2023).
- 226. Goldoni, E.; Savazzi, P.; Favalli, L.; Vizziello, A. Correlation between weather and signal strength in Lorawan Networks: An extensive dataset. *Comput. Netw.* 2022, 202, 108627. [CrossRef]
- 227. Lagat, S.J. Detecting Denial of Service Attacks in LoRaWAN. Master's Thesis, St. Pölten University of Applied Sciences, Pölten, Austria, 2022.
- González-Palacio, M.; Tobón-Vallejo, D.; Sepúlveda-Cano, L.M.; Rúa, S.; Pau, G.; Le, L.B. LoRaWAN Path Loss Measurements in an Urban Scenario including Environmental Effects. *Data* 2022, 8, 4. [CrossRef]
- LoRaWAN Path Loss Dataset. Available online: https://github.com/magonzalezudem/MDPI\_LoRaWAN\_Dataset\_With\_ Environmental\_Variables (accessed on 11 July 2023).
- Ren, Y.; Liu, L.; Li, C.; Cao, Z.; Chen, S. Is LoRaWAN Really Wide? Fine-grained LoRa Link-level Measurement in An Urban Environment. In Proceedings of the 2022 IEEE 30th International Conference on Network Protocols (ICNP), Lexington, KY, USA, 30 October 2022–2 November 2022; pp. 1–12. [CrossRef]
- Yao, Y.; Ma, Z.; Cao, Z. LoSee: Long-Range Shared Bike Communication System Based on LoRaWAN Protocol. In Proceedings of the EWSN, Beijing, China, 25–27 February; pp. 407–412.
- 232. LoSee Dataset. Available online: https://github.com/lilygeek/LoSee\_ICNP (accessed on 9 July 2023).
- 233. Kumar, R.; Mishra, R.; Gupta, H.P. A Federated Learning Approach With Imperfect Labels in LoRa-Based Transportation Systems. *IEEE Trans. Intell. Transp. Syst.* 2023, 1–9. [CrossRef]
- 234. Signal Quality Measurement (SQM) Dataset. 2023. Available online: https://dx.doi.org/10.21227/aysz-nq69 (accessed on 10 July 2023).
- 235. Eldeeb, E.; Alves, H. LoRaWAN-enabled Smart Campus: The Dataset and a People Counter Use Case. *arXiv* 2023, arXiv:2304.13366.
- 236. The Smart Campus Dataset. 2023. Available online: https://dx.doi.org/10.21227/xe4q-ax22 (accessed on 10 July 2023).
- 237. Farhad, A. AI-ERA: Artificial Intelligence-Empowered Resource Allocation for LoRa-Enabled IoT Applications. Available online: https://github.com/afarhad/AI-ERA (accessed on 8 July 2023).
- 238. LoRaWAN Traffic Analysis Dataset. Available online: https://zenodo.org/record/8090619 (accessed on 9 July 2023).
- 239. Outdoor LoRa RSSI Dataset. Available online: https://github.com/oliveiraleo/LoRa-RSSI-dataset-outdoor (accessed on 11 July 2023).
   240. LoRa RSSI Measurements. Available online: https://github.com/oliveiraleo/RSSignal-LoRa (accessed on 11 July 2023).
- 240. Loka KSSi measurements. Avanable online: https://github.com/onverlateo/KSSignal-Loka (accessed on 11 July 2025).
- 241. LoRaWAN Dataset for Spreading Factor. Available online: https://github.com/IbrahimAqeel2023/LoRaWAN-Dataset-Combine-SFs/tree/main (accessed on 11 July 2023).
- 242. LoRa Time Series Dataset. Available online: https://github.com/akapet00/lora-time-series/tree/master (accessed on 11 July 2023).
- 243. ns3-AI Module. Available online: https://github.com/hust-diangroup/ns3-ai (accessed on 18 January 2023).
- 244. Yin, H.; Liu, P.; Liu, K.; Cao, L.; Zhang, L.; Gao, Y.; Hei, X. Ns3-Ai: Fostering Artificial Intelligence Algorithms for Networking Research. In Proceedings of the 2020 Workshop on Ns-3, Gaithersburg, MD, USA, 17–18 June 2020; pp. 57–64. [CrossRef]
- 245. API for Reinforcement Learning. Available online: https://www.gymlibrary.dev/i (accessed on 18 January 2023).
- 246. Gawłowicz, P.; Zubow, A. ns-3 meets OpenAI Gym: The Playground for Machine Learning in Networking Research. In Proceedings of the ACM International Conference on Modeling, Analysis and Simulation of Wireless and Mobile Systems (MSWiM), Miami Beach, FL, USA, 25–29 November 2019.
- 247. Demeslay, C.; Rostaing, P.; Gautier, R. Theoretical Performance of LoRa System in Multi-Path and Interference Channels. *IEEE Internet Things J.* **2022**, *9*, 6830–6843. [CrossRef]
- 248. Open Neural Network Exchange (ONNX). Available online: https://github.com/onnx/onnx (accessed on 28 December 2022).
- 249. ORAN: Ns-3 Module for Open Radio Access Network. Available online: https://github.com/usnistgov/ns3-oran/ (accessed on 1 June 2023).
- Singh, S.K.; Singh, R.; Kumbhani, B. The Evolution of Radio Access Network Towards Open-RAN: Challenges and Opportunities. In Proceedings of the IEEE Wireless Communications and Networking Conference Workshops (WCNCW), Seoul, Republic of Korea, 6–9 April 2020; pp. 1–6. [CrossRef]
- NS3 FL: Federated Learning Simulator. 2020. Available online: https://github.com/eekaireb/ns3-fl-network/tree/ec4276cc8 61f2ff5afdba23571790b8783d56790 (accessed on 7 May 2023).
- 252. Ekaireb, E.; Yu, X.; Ergun, K.; Zhao, Q.; Lee, K.; Huzaifa, M.; Rosing, T. ns3-fl: Simulating Federated Learning with ns-3. In Proceedings of the 2022 Workshop on ns-3, Virtual Event, 22–23 June 2022; pp. 97–104.
- FLSim: A Federated Learning Simulator. 2020. Available online: https://github.com/eekaireb/flsim/tree/a81520201905881f8 8b8940dd6d041ef12a3f1fa (accessed on 31 May 2023).
- Wang, H.; Kaplan, Z.; Niu, D.; Li, B. Optimizing Federated Learning on Non-IID Data with Reinforcement Learning. In Proceedings of the IEEE INFOCOM 2020—IEEE Conference on Computer Communications, Toronto, ON, Canada, 6–9 July 2020; pp. 1698–1707. [CrossRef]
- McMahan, B.; Moore, E.; Ramage, D.; Hampson, S.; y Arcas, B.A. Communication-efficient learning of deep networks from decentralized data. In Proceedings of the Artificial Intelligence and Statistics, Fort Lauderdale, FL, USA, 20–22 April 2017; pp. 1273–1282.
- 256. Chen, Y.; Ning, Y.; Slawski, M.; Rangwala, H. Asynchronous Online Federated Learning for Edge Devices with Non-IID Data. In Proceedings of the 2020 IEEE International Conference on Big Data (Big Data), Atlanta, GA, USA, 10–13 December 2020; pp. 15–24. [CrossRef]

- 257. LoRaWAN Bandits. Available online: https://github.com/renzoe/LoRaWAN-Bandits/tree/main (accessed on 7 July 2023).
- 258. Finnegan, J.; Farrell, R.; Brown, S. Analysis and enhancement of the lorawan adaptive data rate scheme. *IEEE Internet Things J.* **2020**, *7*, 7171–7180. [CrossRef]

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.